# 17 WORKING PAPERS 2020

## CLIMATE CHANGE MITIGATION POLICIES: AGGREGATE AND DISTRIBUTIONAL EFFECTS

Tiago Cavalcanti | Zeina Hasna Cezar Santos



BANCO DE PORTUGAL

# 17 WORKING PAPERS 2020

## CLIMATE CHANGE MITIGATION POLICIES: AGGREGATE AND DISTRIBUTIONAL EFFECTS

Tiago Cavalcanti | Zeina Hasna Cezar Santos

NOVEMBER 2020

The analyses, opinions and findings of these papers represent the views of the authors, they are not necessarily those of the Banco de Portugal or the Eurosystem

Please address correspondence to Banco de Portugal, Economics and Research Department Av. Almirante Reis, 71, 1150-012 Lisboa, Portugal Tel.: +351 213 130 000, email: estudos@bportugal.pt



Lisboa, 2020 • www.bportugal.pt

 Working Papers
 | Lisboa 2020 • Banco de Portugal Av. Almirante Reis, 71 | 1150-012 Lisboa • www.bportugal.pt •

 Edition Economics and Research Department • ISBN (online) 978-989-678-753-0 • ISSN (online) 2182-0422

#### Climate Change Mitigation Policies: Aggregate and Distributional Effects

**Tiago Cavalcanti** University of Cambridge & São Paulo School of Economics–FGV Zeina Hasna University of Cambridge

Cezar Santos Banco de Portugal FGV EPGE & CEPR

November 2020

#### Abstract

We evaluate the aggregate and distributional effects of climate change mitigation policies using a multi-sector equilibrium model with intersectoral inputoutput linkages and worker heterogeneity calibrated to different countries. The introduction of carbon taxes leads to changes in relative prices and inputs reallocation, including labor. For the United States, reaching its Paris Agreement pledge would imply at most a 0.6% drop in output. This impact is distributed asymmetrically across sectors and individuals. Workers with a comparative advantage in dirty energy sectors who do not reallocate bear relatively more of the cost but constitute a small fraction of the labor force.

JEL: E13, H23, J24

Keywords: Climate change, carbon taxes, worker heterogeneity, labor reallocation.

E-mail: tvdvc2@cam.ac.uk; zh274@cam.ac.uk; cezarsantos.econ@gmail.com

Acknowledgements: We thank Jakob Berndt, Tom Berrett, Giancarlo Corsetti, Simon Dietz, Miguel H. Ferreira, John Hassler, Pete Klenow, Per Krusell, Jason Lu, Kamiar Mohaddes, David Newbery, Gustavo Nicolas Paez, Christina Romer, Nisreen Salti, Lidia Smitkova, John Spray, Marina M. Tavares, Richard Tol, Dan Wales, Michael Waugh and participants at numerous workshops for helpful remarks. We gratefully acknowledge financial support from the Keynes Fund at University of Cambridge, the Coordenação de Aperfeiçoamento de Pessoal de Nível Superior—Brasil (CAPES)—Finance Code 001 and Faperj. The views expressed in this article are those of the authors and do not necessarily represent those of Banco de Portugal or the Eurosystem.

#### 1. Introduction

As greenhouse gas emissions reach alarming levels, there is increasing pressure on countries to adopt more aggressive environmental policies. However, concerns regarding their economic and distributional effects hinder the adoption of these policies. Reducing emissions means reallocating resources away from high-carbon sectors towards low-carbon ones. This paper investigates the aggregate and distributional effects of climate change mitigation policies, by focusing on the reallocation of inputs, and labor in particular, across the different sectors of the economy.

We develop a framework that integrates the workers skill distribution with the economys sectoral composition. The theory builds on the Roy model of occupational choice with endogenous human capital investment, in which workers choose sectors based on relative wages and their sector-specific abilities (Roy 1951; Hsieh *et al.* 2019). The model economy consists of various sectors, including four energy-producing activities: oil, coal, natural gas and green. A carbon tax is introduced to the "dirty" energy producers, which in turn affects their prices.<sup>1</sup> Given the intersectoral linkages in the economy, these changes in relative prices create substitution possibilities between all inputs of production, which lead to labor reallocation across sectors. The overall economic impact depends on the magnitude of the tax and on how the revenue is rebated to the economy. As economies differ in their production structures and labor force characteristics, the impact of carbon taxes is likely to vary across countries. We thus calibrate the model parameters for the following six countries: Brazil, Canada, China, India, Mexico and the United States.

We investigate the economic impact of introducing a 32.3% carbon tax, which is the estimated tax needed for the United States to achieve its original Paris Agreement pledge of a 26% reduction in emissions (Ramstein *et al.* 2019). This carbon tax costs the United States at most a 0.6% drop in output, when the government does not rebate its tax revenue. We then implement the same policy for the remaining five countries in order to capture the heterogeneity in responses among these economies. We show that, in the worst case scenario, a 32.3% carbon tax can cause a GDP loss ranging from 0.5% (for Brazil) to 2.1% (for China). However, this drop in GDP can be partially offset by implementing tax rebates.

The relatively small aggregate effects mask sizable heterogeneity at the sectoral and individual levels, where there are non-trivial distributional effects. Our results show that dirty energy sectors exposed to the carbon tax witness the largest drop in wages, and consequently the largest labor outflow. By examining the skill distribution, we show that marginal (relatively less-talented) workers in dirty energy production choose to reallocate away from the taxed sectors. Workers with a strong

<sup>1.</sup> The "dirty energy sectors" hereafter refer to oil, coal and natural gas sectors. "Non-dirty energy sector" refers to the green sector.

comparative advantage in the dirty energy production remain working in this sector and end up bearing the cost of the drop in wages. Nevertheless, these workers constitute a small fraction of the labor force.

An important research agenda has concentrated on finding the optimal level of carbon taxation by integrating the climate and the economy into a single model, as proposed by Nordhaus (1994); Dietz and Stern (2015); Golosov *et al.* (2014); Hassler *et al.* (2018); Tol (2018). Their findings point to the effectiveness of carbon taxes in curbing greenhouse gas emissions and reducing emission-induced economic damage. However, this research abstracts from the distributional impacts of climate change mitigation policies, which is a key feature of our analysis.

There is a related literature investigating the distributional effects of carbon taxation. Grainger and Kolstad (2010), for instance, find that carbon taxes tend to be regressive by focusing on the use-side incidence of such taxes, since lower-income households devote a larger share of their expenditures to energy consumption. However, other papers show that carbon taxes can have progressive impacts once the source-side, i.e. the relative change in remuneration of factor inputs, is taken into account and tax revenues are rebated (e.g., Dissou and Siddiqui (2014); Goulder et al. (2019); Tavares (2020); Chateau et al. (2018); Bosetti and Maffezzoli (2013)). Our paper investigates the aggregate and distributional effects of carbon taxes by exploring how changes in factor prices induced by carbon taxes cascade to the rest of the economy and lead to sectoral reallocation of inputs, including labor.<sup>2</sup> Individuals in our framework are heterogeneous in their productivity to work in different economic activities and choose their human capital investment based on their occupational choice and the human capital return in such occupation. Therefore, carbon taxes induce also endogenous change in human capital investment.

The rest of this paper is structured as follows. Section 2 describes the model and characterizes its equilibrium. Section 3 discusses the calibration strategy. Section 4 presents the aggregate results and Section 5 the sectoral- and individual-level results of the counterfactual analyses. Section 6 concludes.

#### 2. The Model

The model economy consists of individuals who live for two periods. As in Hsieh *et al.* (2019), in the first period of life, workers draw an ability vector that determines their productivity for working in each sector of the economy. They make their occupational choice and invest in human capital. In the second period,

<sup>2.</sup> There are some papers studying the effects of climate policies on jobs by considering the unemployment dimension (e.g., Aubert and Chiroleu-Assouline (2019); Castellanos and Heutel (2019); Hafstead and Williams III (2018)). They are related to the literature covering the labor market dimension of non-climate policies such as trade and innovation policies. See, for instance, Adão *et al.* (2020); David *et al.* (2013); Lyon and Waugh (2019).

individuals work and consume. On the production side, each sector produces a distinct intermediate good, including four types of energy: oil, coal, natural gas and green. There is also a final good sector. We describe the details of the model environment below.

#### 2.1. Households

There is a continuum of measure one of individuals, each working in one of the J intermediate sectors. Individuals have two units of time: one unit when they are "young", which is allocated between leisure and schooling; and one unit when they are "old", when they supply their labor inelastically to one of the intermediate goods sectors.

Each individual derives utility from consumption, c, and leisure, (1 - s), according to the following utility function:<sup>3</sup>

$$U = c^{\gamma}(1-s), \ \gamma > 0,$$

where s denotes time spent on schooling in the first period of life and  $\gamma$  controls the relative weight of consumption in the individual's utility.

Human capital for sector j depends on schooling time, s, and schooling resources (like books or tuition fees), e, and is given by:

$$h(s,e) = s^{\varphi_j} e^{\eta}.$$

The elasticity of human capital with respect to time is sector-specific,  $\varphi_j$ , such that different sectors feature different returns to schooling.

The individual's labor income is the product of the wage per efficiency unit in sector j,  $w_j$ , their idiosyncratic ability draw,  $z_j$ , and their acquired human capital for sector j, h(s, e). Individual income is split between consumption, c, and expenditures on schooling resources, e. Given an occupational choice, wage, and idiosyncratic talent,  $z_j$ , the individual's utility maximization problem is given by:

$$U_j(w_j, z_j) = \max_{c,s,e} c^{\gamma} (1-s) \text{ subject to } c = w_j z_j h(s,e) - e.$$
 (1)

The equations describing the solution of this problem are:

$$s_j^* = \frac{1}{1 + \frac{1-\eta}{\gamma\varphi_j}},\tag{2}$$

$$e_{j}^{*}(z_{j}) = \left(\eta w_{j} z_{j}(s_{j}^{*})^{\varphi_{j}}\right)^{\frac{1}{1-\eta}}.$$
(3)

Note that changes in wages do not affect the individual choices of time spent on schooling  $(\partial s_j^*/\partial w_j = 0)$ , but do affect the amount of goods spent on schooling resources  $(\partial e_j^*/\partial w_j > 0)$ .

<sup>3.</sup> To save on notation, individuals will not be indexed by a superscript.

After substituting equations (2) and (3) into (1), the individual's indirect utility reads:

$$U_{j}^{*} = \left[ w_{j} z_{j} s_{j}^{\varphi_{j}} (1 - s_{j})^{\frac{1 - \eta}{\beta}} \eta^{\eta} (1 - \eta)^{(1 - \eta)} \right]^{\frac{\beta}{1 - \eta}}.$$
 (4)

2.1.1. Occupational Skills. Each worker is endowed with a vector of idiosyncratic abilities  $\{z_j\}_{j=1}^J$ . We assume that the individual's abilities for the J sectors are drawn from a multivariate Fréchet distribution, such that:

$$F(z_1, ..., z_J) = \exp\left(-\sum_{j=1}^J (z_j)^{-\lambda}\right), \ \lambda > 1,$$

where the parameter  $\lambda$  measures the dispersion of individual productivity across sectors. A higher value of  $\lambda$  corresponds to smaller dispersion. When  $\lambda$  is small, workers' abilities are more dispersed, and hence a larger change in wages is needed to get workers to reallocate across sectors. And vice versa.

2.1.2. Occupational Choice. Self-selection is driven by how heterogeneous abilities interact with the endogenous components of an individual's utility in (4). Workers supply their labor to the sector which offers them the highest relative returns given their vector of ability, i.e. highest utility  $\max_i \{U_i\}$ .

Knowing the decision rule behind workers' occupational choice, we can calculate the share of workers in each of the sectors of the economy.

Proposition 1. The share of workers in sector j, denoted by  $q_j$ , is given by:

$$q_j = \frac{\tilde{w}_j^{\lambda}}{\sum_k \tilde{w}_k^{\lambda}} \text{ where } \tilde{w}_j = w_j s_j^{\varphi_j} (1 - s_j)^{\frac{1 - \eta}{\beta}} \text{ for } j \in \{1, ..., J\}.$$
(5)

**Proof:** See Appendix A.1. ■

Each worker's occupation choice is driven by *relative* returns  $\tilde{w}_j z_j$  instead of *absolute* returns  $\tilde{w}_j$ . Using the tractability afforded by the Fréchet distribution, we can write the share of workers in each sector using (5). Having calculated the labor supply for each sector, we can now compute the efficiency units of labor supplied (i.e. effective labor supply) in each sector.

Proposition 2. The effective labor supply for sector j is given by:

$$L_{j}^{s} = (s_{j}^{\varphi_{j}})^{\frac{1}{1-\eta}} (\eta w_{j})^{\frac{\eta}{1-\eta}} q_{j}^{1-\frac{1}{\lambda}\frac{1}{1-\eta}} \Gamma\left(1-\frac{1}{\lambda}\frac{1}{1-\eta}\right) \text{ for } j \in \{1,...,J\},$$
(6)

where  $\Gamma\left(1-\frac{1}{\lambda}\frac{1}{1-\eta}\right)$  is the Gamma function evaluated at the constant  $\frac{1}{\lambda}\frac{1}{1-\eta}$ .

Proof: See Appendix A.2.

Using equations (5) and (6), we can calculate average worker quality in a sector by taking the ratio of efficiency units of labor supplied over the units of labor supplied,  $L_j^s/q_j$ . Average quality is therefore inversely related to the labor share in each sector, which captures a selection effect.

#### 2.2. Production

Recall that there are J intermediate good sectors and one final good sector. We will now describe each of these in turn.

2.2.1. Intermediate Goods. Our production setup is similar to trade models such as Eaton and Kortum (2002). There are J sectors, each producing a differentiated intermediate good. Among these, there are four energy sectors: oil, coal, natural gas, and green. The first three energy sectors are polluting, so we will refer to them as the "dirty" energy sectors, and the fourth sector is the "clean" energy sector. The technology to produce each intermediate good  $j \in \{1, 2, ..., J\}$  is represented by a Cobb-Douglas function with constant returns to scale:

$$Y_j = L_j^{\beta_j} \prod_{k=1}^J x_{jk}^{\nu_{jk}}, \ \beta_j, \nu_{jk} \in [0,1]; \text{ and } \beta_j + \sum_{k=1}^J \nu_{jk} = 1,$$

where  $L_j$  corresponds to effective labor input and  $\beta_j$  is the labor share in sector j. The variable  $x_{jk}$  denotes the quantity of good k used in the production of good j. The parameter  $\nu_{jk}$  determines the relative importance of good k in the production of sector j.<sup>4</sup> The inclusion of intersectoral linkages allows for a more detailed analysis of the general equilibrium effects of adding a carbon tax (Jones 2011; Acemoglu *et al.* 2012; King *et al.* 2019).

The representative firm in the intermediate good sector j chooses labor  $L_j$  and intermediate inputs  $\{x_{jk}\}_{k=1}^{J}$  to maximize:

$$\pi_j = \max_{L_j, x_{jk}} \left\{ P_j L_j^{\beta_j} \prod_{k=1}^J x_{jk}^{\nu_{jk}} - w_j L_j - \sum_{k=1}^J P_k x_{jk} \right\},\tag{7}$$

where  $P_j$  is the price of intermediate good j and  $w_j$  is the wage rate paid in sector j. The solution to this problem implies that inputs are paid according to their marginal products, such that:

$$\beta_j P_j L_j^{\beta_j - 1} \prod_{k=1}^J x_{jk}^{\nu_{jk}} = w_j,$$
$$\nu_{jk} P_j L_j^{\beta_j} x_{jk}^{\nu_{jk} - 1} \prod_{k \neq s} x_{js}^{\nu_{js}} = P_k, \ \forall x_{jk}, \ k \in \{1, 2, ..., J\}.$$

<sup>4.</sup> Golosov *et al.* (2014) estimated the elasticity of substitution between dirty and clean energy sources to be 0.95 based on a metastudy of 47 studies of interfuel substitution (Stern 2012). Therefore, the unitary elasticity of substitution assumed here seems a reasonable simplification.

2.2.2. Final Good. The technology for the final good,  $Y_f$ , is given by a production function that uses differentiated intermediate goods  $\{Y_j^F\}_{j=1}^J$  according to the following aggregator:

$$Y_f = \prod_{j=1} (Y_j^F)^{\sigma_j}, \ \sigma_j \in [0,1) \text{ and } \sum_{j=1}^J \sigma_j = 1.$$

The final good is the numéraire, such that its price  $P_f$  is normalized to 1. The optimization problem of the representative firm in the final good sector is to choose each input  $\{Y_i^F\}_{i=1}^J$  to maximize:

$$\pi_f = \max_{Y_j} \left\{ \prod_{j=1} \left( Y_j^F \right)^{\sigma_j} - \sum_j P_j Y_j^F \right\},\tag{8}$$

and the optimal demand for each input satisfies:

$$Y_j^F = \sigma_j \frac{Y_f}{P_j}, \ \forall j \in \{1, 2, ..., J\}.$$

#### 2.3. Equilibrium

The stationary competitive equilibrium for this economy consists of individual choices  $\{c, s, e\}$ , individual occupational choices, efficiency units of labor input in each sector  $\{L_j\}_{j=1}^J$ , intermediate goods  $\{Y_j\}_{j=1}^J$ , final output  $Y_f$ , wages  $\{w_j\}_{j=1}^J$  and prices of intermediate goods  $\{P_j\}_{j=1}^J$ , such that:

- Individuals maximize their utility, according to equation (1).
- Individuals supply their labor to the sector that provides them with the highest income according to their abilities.
- Firms producing intermediate goods maximize profits, according to equation (7).
- The representative firm of the final good maximizes profits, according to equation (8).
- All markets clear.

#### 2.4. Carbon Taxation

A carbon tax affects the prices of energy inputs, particularly the more polluting types. Therefore, the burden of the tax on the price of each energy type should depend on the carbon intensity of that particular energy type. Following Golosov *et al.* (2014) and Hassler *et al.* (2018), we differentiate between four energy inputs (oil, coal, natural gas and green) according to their carbon content (intensity of carbon emissions to the atmosphere). Denote this content by  $g_j$ , such that  $g_j \in [0, 1]$ . Green energy types (such as wind and solar) are not associated with

any climate externality, so  $g_{green} = 0$ . The carbon tax rate on each energy type is given by  $\tau_j = \tau g_j \ \forall j$  (note that  $\tau_{green} = 0$  since  $g_{green} = 0$ ).

We introduce the carbon tax as a sales tax to each energy type j, such that profits in energy type j, in the presence of such a tax, are given by:

$$\pi_j = (1 - \tau_j) P_j Y_j - w_j L_j - \sum_{k=1}^J P_k x_{jk}.$$

In our simulations, we consider different ways to allocate revenues raised with carbon taxes and adjust the equilibrium conditions accordingly. For instance, in one counterfactual experiment, we consider the use of tax revenues in dirty energy sectors to subsidize the green energy sector. In that experiment, subsidies are designed such that the carbon tax is revenue neutral (i.e.  $\sum_{j=1}^{J} \tau_j P_j Y_j = 0$ ), which implies that subsidized sectors will have a negative  $\tau_j$ .

#### 3. Calibration

This section discusses how we discipline the model parameters in order to investigate the aggregate and distributional effects of climate change mitigation policies. Since these effects are likely to vary across countries due to country-specific characteristics (e.g. production structure and labor force composition), the parameterization of the model is conducted by disciplining the parameters with micro-level data for a sample of six countries spanning a set of developing and advanced economies, namely: Brazil, Canada, China, India, Mexico and United States. We have prior information about some model parameters, such as the importance of each input in the production of intermediate goods. But other parameters are specific to the analysis and we do not have much information about their magnitude. They will be internally estimated to match key moments of the data. Table B.1 in the Appendix lists all the model parameters and divides them into these two groups.

**External Calibration.** To set values for J,  $\beta_j$ , and  $\nu_{jk}$ , we use data from the World Input Output Database (WIOD), which contains national input-output tables, as well as data on sectoral labor force participation rates, labor compensation, and environmental accounts for the countries in our sample. We use data on intersectoral sales to calculate  $\nu_{jk}$  and set  $\beta_j = 1 - \sum_{k=1}^{J} \nu_{jk}$ . We aggregate the 35 sectors in the WIOD into 15 sectors including one aggregate energy sector (see Table B.2 in the Appendix). We then split the aggregate energy sector into oil, coal, natural gas and green energy production based on the energy input mix of each of the intermediate sectors, according to the WIOD environmental accounts on energy use by sector and energy type. We also use the WIOD environmental accounts data on  $CO_2$  emissions by sector and energy type to calculate the effect

of taxes on emissions.  $^{5}$  More details on these parameters are presented in Appendix B.

The sectoral carbon content,  $g_j$ , is based on Golosov *et al.* (2014). The numbers for oil and coal are  $g_{oil} = 0.846$  and  $g_{coal} = 0.716$ . We replicate their methodology and calculate  $g_{gas} = 0.734$  using estimates from Garg *et al.* (2006).

We follow Hsieh *et al.* (2019) to calibrate  $\eta$  and  $\gamma$ .  $\eta$  is equal to the fraction of output spent on education. From the World Development Indicators (WDI), we collect the most recent data on public expenditure on education (as a share of GDP) and normalize it by labor force participation rate to calculate  $\eta$  for each country.<sup>6</sup> To calibrate  $\gamma$ , we take average earnings in sector j,  $\bar{w}_j = w_j[hz_j] = (1-s)^{\frac{-1}{\gamma}} \eta^{\frac{\eta}{1-\eta}} \Gamma(1-\frac{1}{\lambda}\frac{1}{1-\eta})$ . Note that average earnings is proportional to  $(1-s)^{\frac{-1}{\gamma}}$ . We use micro-data from the Integrated Public Use Micro-data Series (IPUMS) for each country in our sample, except China.<sup>7</sup> We then calculate the average years of schooling divided by a pre-work time endowment of 25 years,  $\bar{s}$ , and estimate the Mincerian return to schooling across sectors for each country. With  $\bar{s}$  and  $\xi$ , we calculate  $\gamma = \frac{1}{\xi(1-\bar{s})}$ . The values for  $\eta$  and  $\gamma$  for each country are presented in Tables B.3 and B.5 in the Appendix, respectively.

**Internal Calibration.** The remaining parameters  $\sigma_j$ ,  $\varphi_j$  and  $\lambda$  are disciplined by solving the model and targeting certain data moments. In particular, we calibrate the expenditure shares  $\sigma_j$  such that the sectoral value added shares in the model match those in the data.

We follow the methodology in Hsieh *et al.* (2019) to estimate  $\varphi_j$  and  $\lambda$ . To estimate  $\varphi_j$ , we use data from the WIOD on the number of employees and labor compensation to calculate the average wage in each sector.<sup>8</sup> This yields the relative sectoral wages, which determine the relative values for  $\varphi_j$ . To find the absolute values of  $\varphi_j$ , we take the ratio of the average wages relative to Agriculture. We calculate average schooling in agriculture  $s_{Agri}$  and then use equation (2) to solve for  $\varphi_{Agri}$ . With this, we pin down the remaining  $\varphi_j$  by targeting the ratio of each sectoral wages relative to Agriculture.<sup>9</sup> Data on the relative ratios of sectoral wages and the values for  $\varphi$  for each country are presented in Table B.6 in the Appendix.

<sup>5.</sup> Note that our framework does not model the feedback effects of emissions on the economy. We compute the change in emissions in order to discipline the size of the carbon tax.

<sup>6.</sup> Two remarks: (i) we obtain expenditure on education for China from the Ministry of Finance for the People's Republic of China. (ii) we re-estimate  $\eta$  using data on public and private expenditure on education from OECD for Brazil, Canada, Mexico and United States. The results were similar and are available upon request.

<sup>7.</sup> For China, we use micro-data from the Chinese Household Income Project, 2013.

<sup>8.</sup> For China, data on number of employees is not available; so we use data on the number of people engaged instead.

<sup>9.</sup> Given the lack of information on the individual energy sectors, we target the ratio of average wage in the aggregate energy sector relative to agriculture.

To estimate  $\lambda$ , we use micro-data on individual wages to fit the distribution of residuals from a cross-sectional regression of log income earned on age-industry dummies in a given year for each country. We then match the coefficient of variation of sectoral residual wages. The values of estimated Fréchet parameters, alongside data and model's estimates of the coefficient of variation of wages for each country are presented in Table B.7 in the Appendix.

**Model Fit.** Following the calibration strategy above, we target sectoral value added shares, ratios of relative wages, and the coefficient of variation of wages. Although labor force participation rates are not targeted, model's estimates of sectoral labor force participation shares are highly correlated to their data counterparts (with an average correlation of 0.80 across countries); see Table B.8 in the Appendix. To target labor force participation shares exactly, in a robustness exercise, we add wedges to the wages to capture pre-existing distortions in the labor market in the benchmark model. We run robustness checks for the case of Mexico, which has the worst fit for untargeted labor force participation rates, and show that the main results hold upon adding such wedges; see Appendix D.

#### 4. The Aggregate Effects of Climate Change Mitigation Policies

To investigate how the economy reacts to climate change mitigation policies, we introduce a carbon tax on the "dirty" energy sectors. We consider five different counterfactual policies in which tax revenue is either: (i) wastefully spent ("Wasteful Spending"); (ii) used to subsidize green energy ("Green Subsidy"); (iii) used to subsidize all non-dirty sectors in the economy ("Useful Spending"); (iv) used to subsidize education expenditures for all non-dirty sectors ("Education Subsidy"); or (v) rebated back to households uniformly as lump sum transfers ("Household Transfers").<sup>10</sup>

Similar to King *et al.* (2019), our model does not feature an externality since emissions do not affect production or consumption. Hence, given the absence of externalities, the optimal carbon tax is zero in the benchmark model. Our goal is not to investigate the optimal policy but to understand the aggregate and distributional effects of imposing a carbon tax aimed at reducing emissions. Therefore, our exercises are positive rather than normative.

More specifically, our experiments increase the tax rate on oil, coal and gas energy production sectors from  $\tau = 0\%$  to  $\tau = 32.3\%$ .<sup>11</sup> This is the tax rate needed for the United States to achieve its original Paris Agreement pledge to reduce total emissions by 26% (Ramstein *et al.* 2019). We also apply the 32.3% carbon tax to five different advanced and emerging economies: Brazil, Canada, China, India and

<sup>10.</sup> In policies (ii)-(v), subsidies are designed such that the government budget is balanced.

<sup>11.</sup> Adding a 32.3% value added tax translates into a tax  $\tau_{oil} = 27.3\%$  on oil sales,  $\tau_{coal} = 23.1\%$  on coal sales, and  $\tau_{gas} = 23.7\%$  on gas sales upon adjusting for the carbon content of each energy input.

Mexico. Investigating countries with different levels of development and production structures allows us to capture heterogeneous responses across countries to the same climate change mitigation policy.

Table 1 displays the main aggregate results for this analysis. Panel A reports the results on emissions, GDP, consumption and welfare of introducing a 32.3% carbon tax in the six countries analyzed. For each country, these results are presented for the different types of tax rebates. Take the United States, for instance. By construction, in the wasteful spending scenario, the tax leads to a 26% reduction in total emissions. Since the dirty energy sectors pollute more than the other activities, the drop in their emissions is larger (26.8%). As energy becomes more expensive, the economy contracts and GDP falls by 0.6%. With the tax, reallocation of resources and fall in output, aggregate welfare decreases.

If the government uses the carbon tax revenue to subsidize the green sector, the fall in GDP is dampened to only 0.3%. With more economic activity, emissions actually decline by less than with wasteful spending even with subsidies to the clean sector. An alternative is to subsidize all non-dirty sectors (Useful Spending). Again, the fall in GDP is dampened relative to the wasteful scenario, but emissions do not fall by as much.

When tax revenues are used to finance education subsidies, then US GDP rises by 0.4%. Individuals invest more in education with this policy, increasing individual productivity and therefore aggregate output. This supports the possibility of a "double-dividend" hypothesis, which posits that the carbon tax reduces emissions (environmental dividend) and improves output (economic dividend). Moreover, the education subsidy partially insures individuals, and this can lead to an increase in welfare. This type of insurance coupled with a hike in leisure also explains why welfare increases substantially in the household transfer scenario even though GDP declines. However, given the implausibility of lump-sum transfers in the real world, we do not focus on household transfer results for the remainder of the analysis.

Panel A of Table 1 also displays the results for the other five countries in our sample. The main insights across the different types of tax rebates found for the United States carry over to the other countries. The main differences are on the magnitude of the effects. The reduction in total emissions ranges from 24.0% in Canada to 32.1% in China. With regards to GDP, the losses for the wasteful spending counterfactual range from 0.5% in Brazil to 2.1% in China. The amplitude for the other scenarios is comparable. Two key messages from our results stand out. First, the carbon tax has heterogeneous aggregate effects amongst these six countries. This hinges on the varying importance of the taxed sectors in each country's total value added, intermediate consumption and/or labor force composition. Second, a 32.3% carbon tax seems to have relatively small aggregate output effects. This happens because the dirty energy sectors constitute a small fraction of the gross output in each economy. More detailed results are reported in Tables C.1 and C.2 in the Appendix.

Panel A: 32.3% Carb	on Tax				
Brazil	Total Emissions	Fossil Emissions	GDP	Consumption	Cons. Equiv.
Wasteful Spending	-25.9	-27.5	-0.5	-1.4	-0.9
Green Subsidy	-25.0	-26.5	-0.2	-0.2	-0.2
Useful Spending	-25.5	-27.1	-0.4	-0.4	0.1
Education Subsidy	-25.9	-27.5	0.4	-0.5	0.1
Household Transfers	-25.9	-27.5	-1.1	-1.0	1.3
Canada	Total Emissions	Fossil Emissions	GDP	Consumption	Cons. Equiv.
Wasteful Spending	-24.0	-26.6	-1.2	-3.9	-2.9
Green Subsidy	-22.9	-25.5	-0.8	-0.8	-0.9
Useful Spending	-22.8	-25.4	-1.1	-1.1	0.2
Education Subsidy	-24.0	-26.6	1.2	-1.6	-0.3
Household Transfers	-24.0	-26.6	-3.1	-2.9	2.5
China	Total Emissions	Fossil Emissions	GDP	Consumption	Cons. Equiv.
Wasteful Spending	-32.1	-34.0	-2.1	-6.0	-4.7
Green Subsidy	-25.9	-27.5	-1.2	-1.2	-1.9
Useful Spending	-29.1	-31.0	-1.9	-1.9	-0.4
Education Subsidy	-32.1	-34.0	0.9	-3.1	-1.7
Household Transfers	-32.1	-34.0	-4.5	-4.3	2.8
India	Total Emissions	Fossil Emissions	GDP	Consumption	Cons. Equiv.
Wasteful Spending	-28.2	-29.9	-1.0	-2.9	-2.1
Green Subsidy	-25.4	-26.9	-0.5	-0.5	-0.7
Useful Spending	-26.7	-28.3	-0.8	-0.8	0.0
Education Subsidy	-28.2	-29.9	0.7	-1.2	-0.2
Household Transfers	-28.2	-29.9	-2.7	-2.5	2.5
Mexico	Total Emissions	Fossil Emissions	GDP	Consumption	Cons. Equiv.
Wasteful Spending	-25.3	-26.8	-1.1	-3.4	-2.2
Green Subsidy	-24.6	-26.1	-0.7	-0.7	-0.8
Useful Spending	-24.2	-25.7	-1.0	-1.0	0.4
Education Subsidy	-25.3	-26.8	1.0	-1.4	0.0
Household Transfers	-25.3	-26.8	-3.0	-2.8	2.7
United States	Total Emissions	Fossil Emissions	GDP	Consumption	Cons. Equiv.
Wasteful Spending	-26.0	-26.8	-0.6	-1.7	-1.1
Green Subsidy	-24.3	-25.0	-0.3	-0.3	-0.3
Useful Spending	-25.3	-26.1	-0.5	-0.5	0.1
Education Subsidy	-26.0	-26.8	0.4	-0.7	0.1
Household Transfers	-26.0	-26.8	-1.9	-1.8	1.1
Panel B: 25.4% Carb	on Tax				
China	Total Emissions	Fossil Emissions	GDP	Consumption	Cons. Equiv.
Wasteful Spending	-26.0	-27.5	-1.5	-4.7	-3.6
Green Subsidy	-20.6	-21.8	-0.7	-0.7	-1.2
Useful Spending	-23.4	-24.8	-1.3	-1.3	-0.1
Education Subsidy	-26.0	-27.5	1.0	-2.2	-1.0
Household Transfers	-26.0	-27 5	-3.5	-3.3	25

Table 1. The Effects of Climate Change Mitigation Policy Under All Recycling Schemes

#### 4.1. China versus the United States

Panel A in Table 1 reported the results for a 32.3% carbon tax, which is needed for the United States to achieve its original Paris Agreement goal. Such a policy yields different effects across countries. Instead of applying the same climate policy for all countries, this subsection solves for the tax rate needed for a country to achieve the same climate target as the United States; i.e. a reduction in emissions of 26%. We take China as an example and find that it requires a carbon tax of 25.4% to achieve such a reduction in emissions. Panel B of Table 1 reports the results for this counterfactual.

A similar reduction in emissions in the United States and China generates larger aggregate output effects in China. For the wasteful spending case, the same reduction in emissions leads to a fall in GDP of 0.5% and 1.5% in the United States and China, respectively. This is due to the fact that China is more reliant on dirty energy than the United States. China has a higher share of oil, coal and natural gas in its intermediate consumption (5.4% vs. 1.8%), value added (4.4% vs. 3.0%) and labor force composition (1.9% vs. 0.6%) compared with the United States. See Table C.2 for more details.

#### 5. The Distributional Effects of Climate Change Mitigation Policies

The previous section highlighted the relatively small aggregate losses caused by the introduction of a carbon tax. This section investigates the sectoral- and individual-level effects of this policy.

#### 5.1. Sectoral-level Analysis

Introducing the carbon tax on oil, coal and natural gas energy sectors causes them to downsize as they become more expensive relative to other sectors. As a result, labor demand and wages in these sectors fall. Workers re-optimize their occupational decisions and some switch sectors. Figure 1 shows the changes in equilibrium labor by sectors. Employment in the oil, coal and natural gas sectors drops, while it increases in the non-dirty sectors of the economy. With the subsidy to clean energy, inputs are reallocated from the dirty energy sectors to the green sector to equalize marginal returns. With an education subsidy, human capital rises because education becomes relatively cheaper, reinforcing the increase in effective labor to the sectors not directly affected by the carbon tax.

The occupational decision of workers is driven by their abilities and the wage in each occupation. Marginal workers with relatively low productivity in the dirty energy sectors reallocate to other sectors of the economy. Workers with a high comparative advantage in the dirty energy sectors remain in these sectors after the policy change. Therefore, due to a selection effect, the average productivity of workers in the taxed sectors rises (see Figure 2). In the green subsidy scenario, average productivity drops significantly in the green sector due to the inflow of workers to this sector, as depicted in Figure 1.

#### 5.2. Individual-level Analysis

We now investigate the distributional effects more closely by focusing on individuallevel effects that arise after the introduction of a carbon tax. We split workers into four categories: (i) those who remain in the non-dirty energy sectors; (ii) those who reallocate from non-dirty energy sectors; (iii) those who remain in dirty energy



Figure 1: Percentage Change in Effective Labor Upon Increasing Carbon Tax from 0% (benchmark) to 32.3% Across All Scenarios.

sectors; and (iv) those who reallocate from dirty energy sectors. We then track how their welfare changes after the implementation of the policy.

Table 2 shows that workers who remain in the dirty sectors (oil, coal and gas) experience the largest decline in welfare. Take the United States as an example. In the wasteful spending scenario, the welfare of stayers in the dirty sectors declines by 12.9%. This loss is approximately double the one experienced by those who

14



#### Climate Change Mitigation Policies: Aggregate and Distributional Effects

15

-25

Green Subsidy

Useful Spending

Revenue-Recycling Schemes

Figure 2: Percentage Change in Average Productivity Upon Increasing Carbon Tax from 0% (benchmark) to 32.3% Across All Scenarios.

Education Subsidy

-25

Green Subsidy

Useful Spending

Revenue-Recycling Schemes

Education Subsidy

switch from the dirty sectors (6.8%). Similar numbers are found for the other counterfactuals and countries (see Table Table 2). This decline in welfare is due to the reduction in labor demand and wages in the taxed sectors. The measure of workers directly affected by the introduction of the carbon tax, however, is relatively small—at most 1.9% of the labor force in our sample of countries (see Table C.2).

	Westeful Caradian - Cara Cubrid								
Brazil	Wasteful	Spending	Green	Subsidy	Useful	Spending	Educatio	on Subsidy	
	CE (%)	LFP (%)	CE (%)	LFP (%)	CE (%)	LFP (%)	CE (%)	LFP (%)	
Non-dirty sectors, stayers	-0.9	99.6	0.4	99.7	0.1	99.7	0.1	99.6	
Non-dirty sectors, switchers	-0.9	0.1	2.8	0.1	0.2	0.1	0.1	0.1	
Dirty sectors, stayers	-14.6	0.2	-12.7	0.2	-13.7	0.2	-13.7	0.2	
Dirty sectors, switchers	-7.7	0.1	-6.5	0.1	-6.8	0.1	-6.8	0.1	
Aggregate	-0.9	100.0	-0.2	100.0	0.1	100.0	0.1	100.0	
Canada	Wasteful	Spending	Green	Green Subsidy		Spending	Education Subsidy		
	CE (%)	LFP (%)	CE (%)	LFP (%)	CE (%)	LFP (%)	CE (%)	LFP (%)	
Non-dirty sectors, stayers	-2.6	96.9	0.6	96.4	0.5	96.9	0.0	96.9	
Non-dirty sectors, switchers	-2.5	0.2	9.1	0.7	0.5	0.2	0.1	0.2	
Dirty sectors, stayers	-13.4	2.1	-11.6	2.1	-11.3	2.1	-11.2	2.1	
Dirty sectors, switchers	-7.8	0.7	-5.6	0.8	-5.2	0.8	-5.4	0.7	
Aggregate	-2.9	100.0	-0.9	100.0	0.2	100.0	-0.3	100.0	
China	Wastefu	Spending	Green	Subsidy	Useful	Spending	Educatio	on Subsidy	
	CE (%)	LFP (%)	CE (%)	LFP (%)	CE (%)	LFP (%)	CE (%)	LFP (%)	
Non-dirty sectors, stayers	-4.7	98.2 ´	-0.5	98.Ò	-0.6	98. <b>4</b>	-1.7 ´	98.2 ´	
Non-dirty sectors, switchers	-4.5	0.4	6.0	0.7	-0.4	0.3	-1.4	0.4	
Dirty sectors, stayers	-16.7	0.9	-13.0	1.0	-13.0	0.9	-14.1	0.9	
Dirty sectors, switchers	-10.4	0.4	-6.9	0.4	-6.4	0.4	-7.5	0.4	
Aggregate	-4.7	100.0	-1.9	100.0	-0.4	100.0	-1.7	100.0	
	Wasteful Spending								
India	Wasteful	Spending	Green	Subsidy	Useful :	Spending	Educatio	on Subsidy	
India	Wasteful CE (%)	Spending LFP (%)	Green CE (%)	Subsidy LFP (%)	Useful : CE (%)	Spending LFP (%)	Educatio CE (%)	n Subsidy LFP (%)	
India Non-dirty sectors, stayers	Wasteful CE (%) -2.0	Spending LFP (%) 98.6	Green CE (%) 0.5	Subsidy LFP (%) 98.2	Useful : CE (%) 0.0	Spending LFP (%) 98.6	Educatio CE (%) -0.2	on Subsidy LFP (%) 98.6	
India Non-dirty sectors, stayers Non-dirty sectors, switchers	Wasteful CE (%) -2.0 -1.9	Spending LFP (%) 98.6 0.2	Green CE (%) 0.5 5.9	Subsidy LFP (%) 98.2 0.5	Useful : CE (%) 0.0 0.1	Spending LFP (%) 98.6 0.1	Educatio CE (%) -0.2 -0.1	on Subsidy LFP (%) 98.6 0.2	
India Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, stayers	Wasteful CE (%) -2.0 -1.9 -13.8	Spending LFP (%) 98.6 0.2 0.9	Green CE (%) 0.5 5.9 -11.9	Subsidy LFP (%) 98.2 0.5 0.9	Useful : CE (%) 0.0 0.1 -11.9	Spending LFP (%) 98.6 0.1 0.9	Educatic CE (%) -0.2 -0.1 -12.2	on Subsidy LFP (%) 98.6 0.2 0.9	
India Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, stayers Dirty sectors, switchers	Wasteful CE (%) -2.0 -1.9 -13.8 -7.6	Spending LFP (%) 98.6 0.2 0.9 0.4	Green CE (%) 0.5 5.9 -11.9 -5.9	Subsidy LFP (%) 98.2 0.5 0.9 0.3	Useful : CE (%) 0.0 0.1 -11.9 -5.6	Spending LFP (%) 98.6 0.1 0.9 0.4	Educatic CE (%) -0.2 -0.1 -12.2 -5.9	on Subsidy LFP (%) 98.6 0.2 0.9 0.4	
India Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, stayers Dirty sectors, switchers Aggregate	Wasteful CE (%) -2.0 -1.9 -13.8 -7.6 -2.5	Spending LFP (%) 98.6 0.2 0.9 0.4 100.0	Green CE (%) 0.5 5.9 -11.9 -5.9 -0.7	Subsidy LFP (%) 98.2 0.5 0.9 0.3 100.0	Useful : CE (%) 0.0 0.1 -11.9 -5.6 0.0	Spending LFP (%) 98.6 0.1 0.9 0.4 100.0	Educatic CE (%) -0.2 -0.1 -12.2 -5.9 -0.2	on Subsidy LFP (%) 98.6 0.2 0.9 0.4 100.0	
India Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, stayers Dirty sectors, switchers Aggregate Mexico	Wasteful CE (%) -2.0 -1.9 -13.8 -7.6 -2.5 Wasteful	Spending LFP (%) 98.6 0.2 0.9 0.4 100.0	Green CE (%) 0.5 5.9 -11.9 -5.9 -0.7 Green	Subsidy LFP (%) 98.2 0.5 0.9 0.3 100.0 Subsidy	Useful 2 CE (%) 0.0 0.1 -11.9 -5.6 0.0 Useful 2	Spending LFP (%) 98.6 0.1 0.9 0.4 100.0 Spending	Educatic CE (%) -0.2 -0.1 -12.2 -5.9 -0.2 Educatic	on Subsidy LFP (%) 98.6 0.2 0.9 0.4 100.0 on Subsidy	
India Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, stayers Dirty sectors, switchers Aggregate Mexico	Wasteful CE (%) -2.0 -1.9 -13.8 -7.6 -2.5 Wasteful CE (%)	Spending LFP (%) 98.6 0.2 0.9 0.4 100.0 Spending LFP (%)	Green CE (%) 0.5 5.9 -11.9 -5.9 -0.7 Green CE (%)	Subsidy LFP (%) 98.2 0.5 0.9 0.3 100.0 Subsidy LFP (%)	Useful 2 CE (%) 0.0 0.1 -11.9 -5.6 0.0 Useful 2 CE (%)	Spending LFP (%) 98.6 0.1 0.9 0.4 100.0 Spending LFP (%)	Educatic CE (%) -0.2 -0.1 -12.2 -5.9 -0.2 Educatic CE (%)	on Subsidy LFP (%) 98.6 0.2 0.9 0.4 100.0 on Subsidy LFP (%)	
India Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, stayers Dirty sectors, switchers Aggregate Mexico Non-dirty sectors, stayers	Wasteful CE (%) -2.0 -1.9 -13.8 -7.6 -2.5 Wasteful CE (%) -1.9	Spending LFP (%) 98.6 0.2 0.9 0.4 100.0 Spending LFP (%) 98.6	Green CE (%) 0.5 5.9 -11.9 -5.9 -0.7 Green CE (%) 1.5	Subsidy LFP (%) 98.2 0.5 0.9 0.3 100.0 Subsidy LFP (%) 98.4	Useful : CE (%) 0.0 0.1 -11.9 -5.6 0.0 Useful : CE (%) 0.5	Spending LFP (%) 98.6 0.1 0.9 0.4 100.0 Spending LFP (%) 98.6	Educatic CE (%) -0.2 -0.1 -12.2 -5.9 -0.2 Educatic CE (%) 0.3	n Subsidy LFP (%) 98.6 0.2 0.9 0.4 100.0 n Subsidy LFP (%) 98.6	
India Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, stayers Dirty sectors, switchers Aggregate Mexico Non-dirty sectors, stayers Non-dirty sectors, switchers	Wasteful CE (%) -2.0 -1.9 -13.8 -7.6 -2.5 Wasteful CE (%) -1.9 -1.9	Spending LFP (%) 98.6 0.2 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2	Green CE (%) 0.5 5.9 -11.9 -5.9 -0.7 Green CE (%) 1.5 13.7	Subsidy LFP (%) 98.2 0.5 0.9 0.3 100.0 Subsidy LFP (%) 98.4 0.4	Useful : CE (%) 0.0 0.1 -11.9 -5.6 0.0 Useful : CE (%) 0.5 0.7	Spending LFP (%) 98.6 0.1 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2	Educatic CE (%) -0.2 -0.1 -12.2 -5.9 -0.2 Educatic CE (%) 0.3 0.3	n Subsidy LFP (%) 98.6 0.2 0.9 0.4 100.0 on Subsidy LFP (%) 98.6 0.2	
India Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, stayers Dirty sectors, switchers Aggregate Mexico Non-dirty sectors, stayers Non-dirty sectors, stayers Dirty sectors, stayers	Wasteful CE (%) -2.0 -1.9 -13.8 -7.6 -2.5 Wasteful CE (%) -1.9 -1.9 -1.9 -1.9	Spending LFP (%) 98.6 0.2 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.9	Green CE (%) 0.5 5.9 -11.9 -5.9 -0.7 Green CE (%) 1.5 13.7 -13.1	Subsidy LFP (%) 98.2 0.5 0.9 0.3 100.0 Subsidy LFP (%) 98.4 0.4 0.9	Useful : CE (%) 0.0 0.1 -11.9 -5.6 0.0 Useful : CE (%) 0.5 0.7 -12.8	Spending LFP (%) 98.6 0.1 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.9	Educatic CE (%) -0.2 -0.1 -12.2 -5.9 -0.2 Educatic CE (%) 0.3 0.3 -12.6	nn Subsidy LFP (%) 98.6 0.2 0.9 0.4 100.0 nn Subsidy LFP (%) 98.6 0.2 0.9	
India Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, stayers Dirty sectors, switchers Aggregate Mexico Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, switchers	Wasteful CE (%) -2.0 -1.9 -1.3.8 -7.6 -2.5 Wasteful CE (%) -1.9 -1.9 -14.5 -8.1	Spending LFP (%) 98.6 0.2 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.9 0.3	Green CE (%) 0.5 5.9 -11.9 -5.9 -0.7 Green CE (%) 1.5 13.7 -13.1 -6.6	Subsidy LFP (%) 98.2 0.5 0.9 0.3 100.0 Subsidy LFP (%) 98.4 0.4 0.9 0.3	Useful : CE (%) 0.0 0.1 -11.9 -5.6 0.0 Useful : CE (%) 0.5 0.7 -12.8 -5.9	Spending LFP (%) 98.6 0.1 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.9 0.3	Educatic CE (%) -0.2 -0.1 -12.2 -5.9 -0.2 Educatic CE (%) 0.3 0.3 -12.6 -6.1	nn Subsidy LFP (%) 98.6 0.2 0.9 0.4 100.0 on Subsidy LFP (%) 98.6 0.2 0.9 0.3	
India Non-dirty sectors, stayers Dirty sectors, switchers Dirty sectors, switchers Aggregate Mexico Non-dirty sectors, stayers Dirty sectors, stayers Dirty sectors, stayers Dirty sectors, switchers Aggregate	Wasteful CE (%) -2.0 -1.9 -13.8 -7.6 -2.5 Wasteful CE (%) -1.9 -1.9 -14.5 -8.1 -2.7	Spending           LFP (%)           98.6           0.2           0.9           0.4           100.0           Spending           LFP (%)           98.6           0.2           0.9           0.4           100.0	Green CE (%) 0.5 5.9 -11.9 -5.9 -0.7 Green CE (%) 1.5 13.7 -13.1 -6.6 -0.8	Subsidy LFP (%) 98.2 0.5 0.9 0.3 100.0 Subsidy LFP (%) 98.4 0.4 0.9 0.3 100.0	Useful : CE (%) 0.0 -11.9 -5.6 0.0 Useful : CE (%) 0.5 0.7 -12.8 -5.9 0.4	Spending LFP (%) 98.6 0.1 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.9 0.3 100.0	Educatic CE (%) -0.2 -0.1 -12.2 -5.9 -0.2 Educatic CE (%) 0.3 0.3 -12.6 -6.1 0.0	nn Subsidy LFP (%) 98.6 0.2 0.9 0.4 100.0 on Subsidy LFP (%) 98.6 0.2 0.9 0.3 100.0	
India Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, stayers Dirty sectors, switchers Aggregate Mexico Non-dirty sectors, stayers Dirty sectors, switchers Dirty sectors, switchers Aggregate United States	Wasteful CE (%) -2.0 -1.9 -13.8 -7.6 -2.5 Wasteful CE (%) -1.9 -1.9 -1.9 -1.4.5 -8.1 -2.7 Wasteful Wasteful	Spending LFP (%) 98.6 0.2 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.9 0.3 100.0 Spending	Green CE (%) 0.5 5.9 -11.9 -5.9 -0.7 Green CE (%) 1.5 13.7 -13.1 -6.6 -0.8 Green	Subsidy LFP (%) 98.2 0.5 0.9 0.3 100.0 Subsidy LFP (%) 98.4 0.4 0.9 0.3 100.0 Subsidy	Useful : CE (%) 0.0 -5.6 0.0 Useful : CE (%) 0.5 0.7 -12.8 -5.9 0.4 Useful :	Spending LFP (%) 98.6 0.1 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.9 0.3 100.0 Spending	Educatic CE (%) -0.2 -0.1 -12.2 -5.9 -0.2 Educatic CE (%) 0.3 0.3 0.3 -12.6 -6.1 0.0 Educatic	n Subsidy LFP (%) 98.6 0.2 0.9 0.4 100.0 on Subsidy LFP (%) 98.6 0.2 0.9 0.3 100.0 on Subsidy	
India Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, switchers Aggregate Mexico Non-dirty sectors, stayers Dirty sectors, stayers Dirty sectors, switchers Dirty sectors, switchers Aggregate United States	Wasteful CE (%) -2.0 -1.9 -13.8 -7.6 -2.5 Wasteful CE (%) Wasteful CE (%)	Spending LFP (%) 98.6 0.2 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.9 0.3 100.0 Spending LFP (%)	Green CE (%) 0.5 5.9 -11.9 -5.9 -0.7 Green CE (%) 1.5 13.7 -13.1 -6.6 -0.8 Green CE (%)	Subsidy LFP (%) 98.2 0.5 0.9 0.3 100.0 Subsidy LFP (%) 98.4 0.4 0.9 0.3 100.0 Subsidy LFP (%)	Useful : CE (%) 0.0 0.1 -11.9 -5.6 0.0 Useful : CE (%) Useful : CE (%)	Spending LFP (%) 98.6 0.1 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.9 0.3 100.0 Spending LFP (%)	Educatic CE (%) -0.2 -0.1 -12.2 -5.9 -0.2 Educatic CE (%) 0.3 -12.6 -6.1 0.0 Educatic CE (%)	n Subsidy LFP (%) 98.6 0.2 0.9 0.4 100.0 on Subsidy LFP (%) 98.6 0.2 0.9 0.3 100.0 on Subsidy LFP (%)	
India Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, switchers Aggregate Mexico Non-dirty sectors, switchers Dirty sectors, switchers Dirty sectors, switchers Dirty sectors, switchers Aggregate United States Non-dirty sectors, stayers	Wasteful CE (%) -2.0 -1.9 -1.3.8 -7.6 -2.5 Wasteful CE (%) -1.9 -1.4.5 -8.1 -2.7 Wasteful CE (%) -1.1 Wasteful CE (%) -1.1	Spending LFP (%) 98.6 0.2 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.9 0.3 100.0 Spending LFP (%) 99.4	Green CE (%) 0.5 5.9 -11.9 -5.9 -0.7 Green CE (%) 1.5 13.7 -13.1 -6.6 -0.8 Green CE (%) 1.1	Subsidy LFP (%) 98.2 0.5 0.9 0.3 100.0 Subsidy LFP (%) 98.4 0.4 0.9 0.3 100.0 Subsidy LFP (%) 99.3	Useful : CE (%) 0.0 -11.9 -5.6 0.0 Useful : CE (%) 0.5 0.7 -12.8 -5.9 0.4 Useful : CE (%) 0.4 Useful : CE (%) 0.2	Spending LFP (%) 98.6 0.1 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.9 0.3 1000. Spending LFP (%) 99.4	Educatic CE (%) -0.2 -0.1 -12.2 -5.9 -0.2 Educatic CE (%) 0.3 -12.6 -6.1 0.0 Educatic CE (%) 0.1	nn Subsidy LFP (%) 98.6 0.2 0.9 0.4 100.0 nn Subsidy LFP (%) 98.6 0.2 0.9 0.3 1000.0 nn Subsidy LFP (%) 99.4	
India Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, stayers Dirty sectors, switchers Aggregate Mexico Non-dirty sectors, stayers Dirty sectors, stayers Dirty sectors, stayers Dirty sectors, switchers Aggregate United States Non-dirty sectors, stayers Non-dirty sectors, switchers	Wasteful CE (%) -2.0 -1.9 -1.3.8 -7.6 -2.5 Wasteful CE (%) -1.9 -1.4.5 -8.1 -2.7 Wasteful CE (%) -1.1 -1.0	Spending LFP (%) 98.6 0.2 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.9 0.3 100.0 Spending LFP (%) 99.4 0.1	Green CE (%) 0.5 5.9 -11.9 -0.7 Green CE (%) 1.5 13.7 -13.1 -6.6 -0.8 Green CE (%) 1.1 9.5	Subsidy LFP (%) 98.2 0.5 0.9 0.3 100.0 Subsidy LFP (%) 98.4 0.4 0.9 0.3 100.0 Subsidy LFP (%) 99.3 0.1	Useful : CE (%) 0.0 0.1 -11.9 -5.6 0.0 Useful : CE (%) 0.5 0.7 -12.8 -5.9 0.4 Useful : CE (%) 0.2	Spending LFP (%) 98.6 0.1 0.9 0.4 100.0 Spending LFP (%) 99.6 0.3 100.0 Spending LFP (%) 99.4 0.1	Educatic CE (%) -0.2 -0.1 -12.2 -5.9 -0.2 Educatic CE (%) 0.3 -12.6 -6.1 0.0 Educatic CE (%) 0.1 0.1	nn Subsidy LFP (%) 98.6 0.2 0.9 0.4 100.0 nn Subsidy LFP (%) 98.6 0.2 0.9 0.3 100.0 nn Subsidy LFP (%) 99.4 0.1	
India Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, switchers Aggregate Mexico Non-dirty sectors, stayers Dirty sectors, switchers Dirty sectors, switchers Aggregate United States Non-dirty sectors, stayers Dirty sectors, switchers Dirty sectors, stayers Dirty sectors, sta	Wasteful CE (%) -2.0 -1.9 -1.3.8 -7.6 -2.5 Wasteful CE (%) -1.9 -1.4.5 -8.1 -2.7 Wasteful CE (%) -1.1 -2.7	Spending LFP (%) 98.6 0.2 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.9 0.3 100.0 Spending LFP (%) 99.4 0.1 0.4	Creen CE (%) 0.5 5.9 -1.9 -5.9 -0.7 Creen CE (%) 1.5 T.3.7 -13.1 -6.6 -0.8 Creen CE (%) 1.1 9.5 -11.5	Subsidy LFP (%) 98.2 0.5 0.9 0.3 100.0 Subsidy LFP (%) 98.4 0.4 0.9 0.3 100.0 Subsidy LFP (%) 99.3 0.1 0.4	Useful : CE (%) 0.0 0.1 -11.9 -5.6 0.0 Useful : CE (%) 0.5 0.7 -12.8 -5.9 0.4 Useful : CE (%) 0.2 0.2 0.2 -11.9	Spending LFP (%) 98.6 0.1 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.3 100.0 Spending LFP (%) 99.4 0.1 0.4	Educatic CE (%) -0.2 -5.9 -0.2 Educatic CE (%) 0.3 -12.6 -6.1 0.0 Educatic CE (%) 0.1 0.1 0.1 0.1	nn Subsidy LFP (%) 98.6 0.2 0.9 0.4 100.0 nn Subsidy LFP (%) 98.6 0.2 0.3 100.0 nn Subsidy LFP (%) 99.4 0.1 0.4	
India Non-dirty sectors, stayers Non-dirty sectors, switchers Dirty sectors, stayers Dirty sectors, switchers Aggregate Mexico Non-dirty sectors, stayers Dirty sectors, switchers Dirty sectors, switchers Aggregate United States Non-dirty sectors, stayers Dirty sectors, switchers	Wasteful CE (%) -2.0 -1.9 -1.8 -7.6 -2.5 Wasteful CE (%) -1.9 -14.5 -8.1 -2.7 Wasteful CE (%) -1.1 -1.1 -1.0 -1.29 -6.8	Spending LFP (%) 98.6 0.2 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.9 0.3 100.0 Spending LFP (%) 99.4 0.1 0.1	CF (%) 0.5 5.9 -11.9 -0.7 CF (%) 1.5 13.7 -13.1 -6.6 -0.8 Green CE (%) 1.1 9.5 -11.5 -5.7	Subsidy LFP (%) 98.2 0.5 0.9 0.3 100.0 Subsidy LFP (%) 98.4 0.4 0.9 0.3 100.0 Subsidy LFP (%) 99.3 0.1 0.4 0.1	Useful : CE (%) 0.0 0.1 -11.9 -5.6 0.0 Useful : CE (%) 0.5 0.7 -12.8 -5.9 0.7 -12.8 -5.9 0.7 Useful : CE (%) 0.2 0.2 -11.9 -5.7	Spending LFP (%) 98.6 0.1 0.9 0.4 100.0 Spending LFP (%) 98.6 0.2 0.9 0.3 100.0 Spending LFP (%) 99.4 0.1 0.1	Educatic CE (%) -0.2 -0.1 -12.2 -5.9 -0.2 Educatic CE (%) 0.3 -12.6 -6.1 0.3 -12.6 -6.1 0.1 0.1 -11.9 -5.7	nn Subsidy LFP (%) 98.6 0.2 0.9 0.4 100.0 nn Subsidy LFP (%) 98.6 0.2 0.9 0.3 100.0 nn Subsidy LFP (%) 99.4 0.1 0.1	

Table 2. Detailed Welfare Analysis by Country

Due to general equilibrium effects, labor reallocation also takes place in the nondirty sectors. Workers from the non-dirty sectors experience welfare gains in most counterfactuals (other than the wasteful spending scenario). This gain is especially large for those workers who switch sectors.

Figure 3 shows the welfare effects of the carbon tax policy across income deciles. The carbon tax has a progressive impact, as it affects more negatively agents at the top deciles of the income distribution. Such qualitative shape is independent of the type of tax rebate considered in our analysis. For most countries, the carbon tax generates welfare gains for almost all workers under the useful spending and education subsidy scenarios. Only workers in the top decile experience welfare losses under these two revenue-recycling schemes. The only exception is China where there are welfare losses across the entire income distribution. This is due to fact that China is the most energy-intensive economy in the sample.



Figure 3: Consumption Equivalents per Decile.

#### 6. Concluding Remarks

This paper quantifies the aggregate and distributional effects of climate change mitigation policies within and across countries. Our results for the United States show that, to achieve its original Paris Agreement Goal, a carbon tax of 32.3% is needed and it causes a drop in GDP of at most 0.6%. Applying the same climate policy to other countries in our sample (Brazil, Canada, China, India and Mexico) yields drops in output ranging from 0.5% (Brazil) to 2.1% (China). The heterogeneity in the results is due to varying degrees of importance of the taxed energy sectors in the respective economies in terms of value added, intermediate consumption and labor force shares. These adverse effects on GDP can be partially or entirely offset through tax rebates.

Despite the small effects on output from imposing a carbon tax, there is significant heterogeneity at the sectoral level. The dirty energy sectors exposed to the carbon tax witness the largest drop in wages, and consequently the largest labor outflow. However, by examining the skill distribution, we find that less-talented workers in dirty energy production reallocate away from the taxed sectors into other sectors in the economy, while workers with a comparative advantage in dirty energy production remain and end up bearing most of the cost from the drop in wages. These workers, however, constitute a small fraction of total employment.

Despite featuring considerable heterogeneity in terms of production structure and worker heterogeneity, our model abstracted from tackling some margins. For instance, interesting insights can come from integrating our framework with a richer description from the demand side. Individuals with different income levels are likely to consume distinct baskets of goods. Another possibility is to study the entire time path of adjustment in response to climate policies. We leave these issues for future research.

#### References

- Acemoglu, Daron, Vasco M Carvalho, Asuman Ozdaglar, and Alireza Tahbaz-Salehi (2012). "The network origins of aggregate fluctuations." *Econometrica*, 80(5), 1977–2016.
- Adão, Rodrigo, Martin Beraja, and Nitya Pandalai-Nayar (2020). "Technological Transitions with Skill Heterogeneity Across Generations." Tech. rep., National Bureau of Economic Research.
- Aubert, Diane and Mireille Chiroleu-Assouline (2019). "Environmental tax reform and income distribution with imperfect heterogeneous labour markets." *European Economic Review*, 116, 60–82.
- Bosetti, Valentina and Marco Maffezzoli (2013). "Taxing carbon under market incompleteness."
- Castellanos, Kenneth A and Garth Heutel (2019). "Unemployment, labor mobility, and climate policy." Tech. rep., National Bureau of Economic Research.

- Chateau, Jean, Ruben Bibas, and Elisa Lanzi (2018). "Impacts of green growth policies on labour markets and wage income distribution: a general equilibrium application to climate and energy policies."
- David, H, David Dorn, and Gordon H Hanson (2013). "The China syndrome: Local labor market effects of import competition in the United States." *American Economic Review*, 103(6), 2121–68.
- Dietz, Simon and Nicholas Stern (2015). "Endogenous growth, convexity of damage and climate risk: how Nordhaus' framework supports deep cuts in carbon emissions." *The Economic Journal*, 125(583), 574–620.
- Dissou, Yazid and Muhammad Shahid Siddiqui (2014). "Can carbon taxes be progressive?" *Energy Economics*, 42, 88–100.
- Eaton, Jonathan and Samuel Kortum (2002). "Technology, geography, and trade." *Econometrica*, 70(5), 1741–1779.
- Garg, A, K Kazunari, and T Pulles (2006). "IPCC guidelines for national greenhouse gas inventories. Volume 2: energy. Chapter 1: introduction." Available from Intergovernmental Panel on Climate Change website: http://www.ipcc-nggip.iges.or.jp/public/2006gl/vol2.html.
- Golosov, Mikhail, John Hassler, Per Krusell, and Aleh Tsyvinski (2014). "Optimal taxes on fossil fuel in general equilibrium." *Econometrica*, 82(1), 41–88.
- Goulder, Lawrence H, Marc AC Hafstead, GyuRim Kim, and Xianling Long (2019). "Impacts of a carbon tax across US household income groups: What are the equity-efficiency trade-offs?" *Journal of Public Economics*, 175, 44–64.
- Grainger, Corbett A and Charles D Kolstad (2010). "Who pays a price on carbon?" *Environmental and Resource Economics*, 46(3), 359–376.
- Hafstead, Marc AC and Roberton C Williams III (2018). "Unemployment and environmental regulation in general equilibrium." *Journal of Public Economics*, 160, 50–65.
- Hassler, John, Per Krusell, and Conny Olovsson (2018). "The consequences of uncertainty: climate sensitivity and economic sensitivity to the climate." *Annual Review of Economics*, 10, 189–205.
- Hsieh, Chang-Tai, Erik Hurst, Charles I Jones, and Peter J Klenow (2019). "The allocation of talent and US economic growth." *Econometrica*, 87(5), 1439–1474.
- Jones, Charles I (2011). "Misallocation, economic growth, and input-output economics." Tech. rep., National bureau of economic research.
- King, Maia, Bassel Tarbush, and Alexander Teytelboym (2019). "Targeted carbon tax reforms." *European Economic Review*, 119, 526–547.
- Lyon, Spencer and Michael E Waugh (2019). "Quantifying the losses from international trade." *Unpublished manuscript*.
- Minnesota Population Center. (2019). "Integrated Public Use Microdata Series, International: Version 7.2. [Dataset]." Minneapolis, MN: IPUMS.
- Nordhaus, William D (1994). *Managing the global commons: the economics of climate change*, vol. 31. MIT press Cambridge, MA.
- Ramstein, Celine, Goran Dominioni, Sanaz Ettehad, Long Lam, Maurice Quant, Jialiang Zhang, Louis Mark, Sam Nierop, Tom Berg, Paige Leuschner, et al.

(2019). State and Trends of Carbon Pricing 2019. The World Bank.

- Roy, Andrew Donald (1951). "Some thoughts on the distribution of earnings." *Oxford Economic Papers*, 3(2), 135–146.
- Stern, David I (2012). "Interfuel substitution: a meta-analysis." *Journal of Economic Surveys*, 26(2), 307–331.
- Tavares, Marina (2020). "Carbon Pricing Winners and Losers: Workers, Consumers, and Policy Options." *Forthcoming*.
- Tol, Richard SJ (2018). "The economic impacts of climate change." *Review of Environmental Economics and Policy*, 12(1), 4–25.

### Appendix

#### Appendix A: Appendix – Theory

#### A.1. Proposition 1 - Occupational Shares

The fraction of workers choosing to work in sector j is denoted by  $q_j$ . For simplicity, we present below the fraction of people who choose to work in sector 1 and this calculation procedure can be replicated to all sectors WLOG.

$$\begin{split} q_1 &= Pr(\tilde{w_1}z_1 > \tilde{w_j}z_j \ \forall \ j \neq 1) \\ &= \left[ Pr(\tilde{w_1}z_1 > \tilde{w_j}z_j \ \forall \ j \neq 1 \ | \ z_1) \right] \\ &= \left[ Pr(z_j < \frac{\tilde{w_1}z_1}{\tilde{w_j}} \ \forall \ j \neq 1 \ | \ z_1) \right] \\ &= \int f(z_1)F(\beta_2 z_1)F(\beta_3 z_1)..F(\beta_j z_1)..F(\beta_J z_1)dz_1 \ where \ \beta_j = \frac{\tilde{w_1}}{\tilde{w_j}} \end{split}$$

This is as if we are taking the derivative with respect to the first argument of the new joint distribution  $F(z_1, \beta_2 z_1, ..., \beta_J z_1)$ , so  $q_1$  is now:

$$q_{1} = \int_{0}^{\infty} F_{1}(z_{1}, \beta_{2}z_{1}, \beta_{3}z_{1}, ..., \beta_{j}z_{1}, ..., \beta_{J}z_{1})dz_{1}$$
  
= 
$$\int_{0}^{\infty} F_{1}(z, \beta_{2}z, \beta_{3}z, ..., \beta_{j}z, ..., \beta_{J}z)dz$$

We know that  $F(z_1, z_2, .. z_J) = \exp\left(-\sum_{j=1}^J z_j^{-\lambda}\right)$ , so:

$$F(z,\beta_2 z,...,\beta_J z) = \exp\left(-\sum_{j=1}^J \beta_j^{-\lambda} z^{-\lambda}\right)$$
$$F_1(z,\beta_2 z,...,\beta_J z) = (-)(-\lambda)z^{-\lambda-1}\exp\left(-\sum_{j=1}^J \beta_j^{-\lambda} z^{-\lambda}\right)$$
$$F_1(z,\beta_2 z,...,\beta_J z) = \lambda z^{-\lambda-1}\exp\left(-\sum_{j=1}^J \beta_j^{-\lambda} z^{-\lambda}\right)$$

 $q_1$  is now:

$$q_{1} = \int_{0}^{\infty} \lambda z^{-\lambda-1} \exp\left(-\sum_{j=1}^{J} \beta_{j}^{-\lambda} z^{-\lambda}\right) dz$$
$$= \int_{0}^{\infty} \lambda z^{-\lambda-1} \exp\left(-z^{-\lambda} \bar{\beta}\right) dz, \text{ where } \bar{\beta} = \sum_{j=1}^{J} \beta_{j}^{-\lambda}$$
$$= \int_{0}^{\infty} \lambda z^{-\lambda-1} \exp\left(-\left[(\bar{\beta})^{\frac{-1}{\lambda}} z\right]^{-\lambda}\right) dz$$

We proceed with integration by change of variables, let  $z' = (\bar{\beta})^{\frac{-1}{\lambda}} z$ , then  $dz' = (\bar{\beta})^{\frac{-1}{\lambda}} dz$ , so if we replace z with z':

$$q_{1} = \int_{0}^{\infty} \lambda \left[ \left( \bar{\beta} \right)^{\frac{1}{\lambda}} z' \right]^{-\lambda - 1} \exp \left( - (z')^{-\lambda} \right) \left( \bar{\beta} \right)^{\frac{1}{\lambda}} dz'$$

$$= \left( \bar{\beta} \right)^{-1} \int_{0}^{\infty} \lambda (z')^{-\lambda - 1} \exp \left( - (z')^{-\lambda} \right) dz'$$

$$= \left( \bar{\beta} \right)^{-1} \int_{0}^{\infty} dF(z')$$

$$= \left( \bar{\beta} \right)^{-1}$$

$$= \frac{1}{\bar{\beta}}$$

$$= \frac{1}{\sum_{j=1}^{J} \left( \frac{\bar{w}_{1} - \lambda}{\bar{w}_{j} - \lambda} \right)}$$

$$= \frac{\bar{w}_{1}^{\lambda}}{\sum_{j=1}^{J} \bar{w}_{j}^{\lambda}}$$

More generally:

$$q_j = \frac{\tilde{w_j}^{\lambda}}{\sum_{k=1}^J \tilde{w_k}^{\lambda}} \text{ for } j \in \{1, ..., J\}$$

 $q_{j}% \left( \mathbf{r}_{j}\right) = \left( \mathbf{r}_{j}\right) \left( \mathbf{$ 

#### A.2. Proposition 2: Effective Labor

Following Hsieh *et al.* (2019), the total efficiency units in each occupation (including both talent and human capital accumulation) is number of workers in every sector, given by  $q_j$ , multiplied by average individual productivity given by  $[h_j z_j]$ .

Recall that  $h_j$  is given by:

$$h_j(s_j, e_j) = s_j^{\varphi_j} e_j^{\eta}$$

$$h_j = s_j^{\varphi_j} \left[ \frac{1}{\eta w_j z_j s_j^{\varphi_j}} \right]^{\frac{\eta}{\eta - 1}}$$

$$h_j = (s_j^{\varphi_j})^{1 - \frac{\eta}{\eta - 1}} (\eta w_j z_j)^{\frac{\eta}{1 - \eta}}$$

$$h_j = (s_j^{\varphi_j})^{\frac{1}{1 - \eta}} (\eta w_j)^{\frac{\eta}{1 - \eta}} z_j^{\frac{\eta}{1 - \eta}}$$

 $h_j z_j$  is now given by:

$$\begin{split} h_j z_j &= (s_j^{\varphi_j})^{\frac{1}{1-\eta}} \left(\eta w_j\right)^{\frac{\eta}{1-\eta}} z_j^{\frac{\eta}{1-\eta}} \cdot z_j \\ h_j z_j &= (s_j^{\varphi_j})^{\frac{1}{1-\eta}} \left(\eta w_j\right)^{\frac{\eta}{1-\eta}} z_j^{\frac{1}{1-\eta}} \\ [h_j z_j] &= (s_j^{\varphi_j})^{\frac{1}{1-\eta}} \left(\eta w_j\right)^{\frac{\eta}{1-\eta}} [z_j^{\frac{1}{1-\eta}} | \text{person works in sector j}] \end{split}$$

Next step is to calculate  $[z_j^{\frac{1}{1-\eta}}|$  person works in sector j]. In order to do so, we first calculate  $[z_j^x|$  person works in sector j], where x is some positive exponent.

As shown during the derivations of the occupational share, the new conditional distribution of individual ability  $z_j$  given that the worker sorts into sector j, is given by  $G(z) = F(\beta_1 z, \beta_2 z, ..., \beta_J z) = \exp\left(-\sum_{j=1}^J \beta_j^{-\lambda} z^{-\lambda}\right) = \exp\left(-\frac{1}{q_j} z^{-\lambda}\right)$ . So the conditional distribution, G(z), is itself Fréchet distributed, with a scaling parameter of  $q_j$ . As such, we now have the following:

$$\begin{split} [z_j^x] &= \int_0^\infty z^x dG(z) \\ &= \int_0^\infty \lambda \frac{1}{q_j} z^{-\lambda - 1 + x} e^{-\frac{1}{q_j} z^{-\lambda}} dz \end{split}$$

Using change of variables,  $y = \frac{1}{q_j} z^{-\lambda}$ , we show that:

$$\begin{split} [z_j^x] &= \frac{1}{q_j}^{\frac{x}{\lambda}} \int_0^\infty y^{-\frac{x}{\lambda}} e^{-y} dy \\ &= \frac{1}{q_j}^{\frac{x}{\lambda}} \Gamma(1 - \frac{x}{\lambda}) \\ &= q_j^{-\frac{x}{\lambda}} \Gamma(1 - \frac{x}{\lambda}) \end{split}$$

So for  $x = \frac{1}{1-\eta}$ , we have:

$$\begin{split} [z_j^{\frac{1}{1-\eta}}|\text{person works in sector j}] &= \frac{1}{q_j}^{\frac{1}{\lambda}\frac{1}{1-\eta}}\Gamma(1-\frac{1}{\lambda}\frac{1}{1-\eta})\\ &= q_j^{-\frac{1}{\lambda}\frac{1}{1-\eta}}\Gamma(1-\frac{1}{\lambda}\frac{1}{1-\eta}) \end{split}$$

So average productivity is now:

Avg Productivity<sub>j</sub> = 
$$[h_j z_j] = (s_j^{\varphi_j})^{\frac{1}{1-\eta}} (\eta w_j)^{\frac{\eta}{1-\eta}} q_j^{-\frac{1}{\lambda}\frac{1}{1-\eta}} \Gamma(1 - \frac{1}{\lambda}\frac{1}{1-\eta})$$
 (A.1)

Note that  $L_j^{supply}$  is sum of all individual productivities (i.e. total efficiency units of labor) employed in sector j, which is given by  $L_j^{supply} = q_j \cdot [h_j z_j]$ .

$$L_{j}^{supply} = (s^{\varphi_{j}})^{\frac{1}{1-\eta}} (\eta w_{j})^{\frac{\eta}{1-\eta}} q_{j}^{1-\frac{1}{\lambda}\frac{1}{1-\eta}} \Gamma(1-\frac{1}{\lambda}\frac{1}{1-\eta})$$
(A.2)

To investigate the partial derivative of  $L_j$  with respect to earnings  $\tilde{w_j}$ , we expand  $q_j$  in the equation of effective labor supply and get the following:

$$L_j^e = (s^{\varphi_j})^{\frac{1}{1-\eta}} \left(\eta w_j\right)^{\frac{\eta}{1-\eta}} \left(\frac{\tilde{w_j}^{\lambda}}{\sum_{k=1}^J \tilde{w_k}^{\lambda}}\right)^{1-\frac{1}{\lambda}\frac{1}{1-\eta}} \Gamma(1-\frac{1}{\lambda}\frac{1}{1-\eta})$$

#### Appendix B: Details on Calibration

Table B.1 lists all the model's parameters and classifies them according to the required calibration procedure.

	Externally Calibrated Parameters	Data Source
J	number of sectors	WIOD data
$\nu_{jk}$	input-output shares	WIOD data
$\beta_i^L$	labor shares	WIOD data
$g_{oil} = 84.6\%$	carbon intensity of oil	Golosov et al. (2014)
$g_{coal} = 71.6\%$	carbon intensity of coal	Golosov et al. (2014)
$g_{natural gas} = 73.4\%$	carbon intensity of natural gas	IPCC (2006)
$g_{green} = 0\%$	carbon intensity of green	Golosov et al. (2014)
$\gamma$	consumption weight in the utility function	Mincerian estimate using IPUMS data
η	expenditure on education (% of GDP)	World Development Indicators
	Internally Calibrated Parameters	Moment(s) Targeted
$\sigma_i$	expenditure shares in final good	Sectoral value added from WIOD data
$\tilde{\varphi_i}$	returns of schooling in sector j	Average relative wages using WIOD data
$\lambda$	Fréchet dispersion parameter	Coefficient of variation in earnings from IPUMS data

Table B.1. List of Parameters

The calibration relies on two major data sources: World Input Output Database (WIOD) and the Integrated Public Use Micro-data Series (IPUMS). Both databases present the sectors according to the International Standard Industrial Classification (ISIC) of all economic activities developed by the United Nations, however IPUMS conforms to a top level aggregation of 15 intermediate goods sectors, which we will refer to when aggregating the data of the 35 sectors in the WIOD inputoutput tables. In order to do so, we first collapse the 35 sectors in the WIOD tables to the top-level ISIC rev 4 classification as presented in the first column of Table B.2. Second, we collapse the 21 sectors into the 15 sectors presented in IPUMS databases. Additionally, since the focus of this paper is on taxing dirty energy producing sectors in the economy, we create an aggregate energy sector by merging 'Mining and Quarrying' and 'Electricity' sectors; the sectoral breakdown is now represented in the second column of Table B.2. Third, we split the aggregate energy sector into four energy producing sectors: oil, coal, natural gas and green according to the WIOD environmental accounts on gross energy use by sector and energy commodity. As such we end up with 18 intermediate goods sectors (J=18), which are presented in the third column of Table B.2.

	Sectors (J=21) ISIC Rev4: Top-level Aggregation		Sectors (J=15) IPUMS Aggregation		Sectors (J=18) Authors' Aggregation
A B C D E F G H I J K L M N O P Q R S T U	Agriculture, hunting, forestry and fishing Mining and Quarrying Manufacturing Electricity, gas, steam and air conditioning supply Water supply: severage, waste management and remediation activities Construction Wholesale and retail trade; repair of motor vehicles and motorcycles Transportation and storage Accommodation and food service activities Information and communication Financial and insurance activities Real estate activities Real estate activities Professional, scientific and technical activities Administrative and support service activities Public administration and defence; compulsory social security Education Human health and social work activities Arts, entertainment and recreation Other service activities - and services-producing activities of households for own use Activities of households activities foods - and services-producing activities of households for own use	A C E F G H,J I K L,M,N Q R,S,U T B,D	Agriculture, hunting, forestry and fishing Manufacturing Water supply Construction Wholesale and retail trade Transport, storage and communications Accommodation and food service activities Financial and insurance activities Real estate, renting and business activities Public administration and defence Education Health and social work Arts and other service activities Private household services Total Energy	1. 2. 3. 4. 5. 6. 7. 8. 9. 10. 11. 12. 13. 14. 15. 16. 17. 18.	Agriculture, hunting, forestry and fishing Manufacturing Water supply Construction Wholesale and retail trade Transport, storage and communications Accommodation and food service activities Financial and insurance activities Real estate, renting and business activities Public administration and defence Education Health and social work Arts and other service activities Private household services Oil Energy Production Caol Energy Production Green Energy Production

Table B.2. Intermediate Goods Sectors

Now that the input-output table is aggregated into J = 18, we calculate the input output matrix  $\nu$  which represents intersectoral elasticise, such that each entry  $\nu_{jk}$ :

$$\nu_{jk} = \frac{\text{Input of sector k into sector j}}{\text{Sales of sector j}}$$

As already discussed,  $\beta_j^L$  is calculated by adhering to the constant-returns-to-scale characteristic of our production function, such that  $\beta_j^L + \sum_{k=1}^J \nu_{jk} = 1$ . These parameters are not presented in the Appendix for space purposes.

Table B.3 presents the values for  $\eta$ , which represents the fraction of output spent on education in every country. The public expenditure series is obtained from World Development Indicators. We also refer to OECD for private and public expenditure on education, but that is only available for four countries in our sample.

	Public expenditure on education (%GDP)	Labor Force Participation	η	Public and private expenditure on education (%GDP)	η
Brazil	6.24	63.88	0.10	5.69	0.09
Canada	5.27	65.07	0.08	5.95	0.09
China	3.52	67.99	0.05	-	-
India	3.84	49.29	0.08	-	-
Mexico	4.91	60.68	0.08	5.90	0.10
United States	4.96	62.05	0.08	6.09	0.10

Table B.3. Expenditure on Education in Every Country

As for the internally calibrated estimates, Table B.4 presents the final expenditure shares of each intermediate good alongside the value added shares of each of the intermediate good sectors in the model and in the data. Table B.5 presents the estimated weight of consumption in utility  $\gamma$ , in every country. Table B.6 demonstrates the sector-specific elasticity of human capital accumulation

to schooling years in each country. Finally, Table B.7 presents the Fréchet Parameter and variation coefficient of wages for each country in our sample.

Sector	Braz	il	Cana	da	Chin	a	Indi	a	Mexi	со	United S	states
	$VA_j$ (%)	$\sigma_j$	$ VA_j(\%) $	$\sigma_j$	$VA_j$ (%)	$\sigma_j$	$VA_j$ (%)	$\sigma_j$	VA <sub>j</sub> (%)	$\sigma_j$	$ $ VA $_j$ (%)	$\sigma_j$
1. Agriculture, hunting, forestry and fishing	5.2%	0.038	1.6%	0.018	9.4%	0.036	14.8%	0.111	3.3%	0.016	1.2%	0.005
2. Manufacturing	14.6%	0.222	11.6%	0.087	30.1%	0.329	16.6%	0.293	18.8%	0.228	12.4%	0.131
3. Water supply	0.7%	0.004	0.3%	0.000	0.3%	0.002	0.2%	0.002	0.4%	0.001	0.3%	0.000
4. Construction	6.7%	0.107	7.7%	0.141	6.8%	0.261	7.2%	0.137	7.6%	0.125	3.8%	0.055
<ol><li>Wholesale and retail trade</li></ol>	12.4%	0.094	10.5%	0.106	9.7%	0.048	17.1%	0.087	16.8%	0.141	12.2%	0.131
<ol><li>Transport, storage and communications</li></ol>	8.0%	0.057	8.0%	0.060	7.2%	0.036	11.6%	0.110	7.9%	0.105	9.1%	0.079
<ol><li>Hotels and restaurants</li></ol>	2.4%	0.031	2.1%	0.027	1.9%	0.021	1.4%	0.023	2.2%	0.027	2.8%	0.039
8. Financial services and insurance	6.3%	0.044	5.5%	0.042	6.0%	0.013	5.5%	0.020	3.5%	0.029	7.0%	0.055
9. Public administration and defense	16.6%	0.089	19.5%	0.142	9.7%	0.061	7.7%	0.071	17.8%	0.105	23.1%	0.149
10. Real estate, renting and business activities	9.6%	0.125	9.0%	0.150	4.0%	0.067	6.7%	0.063	4.4%	0.064	13.1%	0.179
11. Education	5.5%	0.066	5.4%	0.063	3.3%	0.050	4.0%	0.040	4.3%	0.049	1.1%	0.016
12. Health and social work	4.2%	0.060	6.3%	0.063	1.8%	0.049	1.5%	0.021	2.3%	0.034	7.1%	0.116
<ol><li>Other services activities</li></ol>	1.8%	0.027	2.1%	0.025	2.3%	0.021	2.4%	0.021	1.6%	0.020	2.6%	0.031
<ol><li>Private households services</li></ol>	1.1%	0.011	0.0%	0.000	0.0%	0.000	0.0%	0.000	0.5%	0.005	0.1%	0.001
15. Oil energy production	1.7%	0.013	4.9%	0.025	1.2%	0.006	0.6%	0.000	4.3%	0.024	1.3%	0.002
16. Coal energy production	0.1%	0.000	0.5%	0.006	2.8%	0.000	0.9%	0.000	0.3%	0.004	0.7%	0.007
<ol><li>Natural gas energy production</li></ol>	0.3%	0.000	2.1%	0.018	0.3%	0.000	0.1%	0.000	2.0%	0.014	0.9%	0.000
18. Green energy production	2.9%	0.011	2.9%	0.028	3.2%	0.000	1.6%	0.000	2.0%	0.011	1.2%	0.002

Table B.4. Intermediate Goods Sectors: Value-Added and Final Expenditure Shares

Country	Consumption Weight ( $\gamma$ )
Brazil	0.3258
Canada	0.4060
China	0.6644
India	0.3809
Mexico	0.5006
United States	0.3176

Table B.5. Consumption Weight in Utility Function for all Countries in the Sample

Sector	Bra	zil	Cana	ada	Chi	na	Ind	ia	Mex	ico	U	S
	$\frac{w_j}{w_{Agri}}$	$\varphi_j$										
1. Agriculture, hunting, forestry and fishing	1.0	0.59	1.0	1.82	1.0	0.6	1.0	0.61	1.0	0.55	1.0	2.81
2. Manufacturing	2.1	0.86	1.0	0.84	1.7	0.8	1.3	0.78	2.5	0.71	1.4	1.89
3. Water supply	1.9	2.49	1.0	2.98	10.1	58.4	4.2	8.19	2.7	5.32	1.3	4.98
4. Construction	1.0	0.52	1.2	1.24	1.4	1.2	1.2	1.00	2.6	1.29	1.5	2.79
5. Wholesale and retail trade	1.2	0.48	0.7	0.52	0.9	0.5	5.7	3.07	3.5	1.24	0.9	1.33
6. Transport, storage and communications	2.7	1.39	1.0	1.04	2.5	2.7	2.3	1.63	4.2	2.27	1.5	2.13
7. Hotels and restaurants	0.9	0.75	0.4	0.56	1.7	3.1	2.0	2.69	2.7	2.55	0.5	1.17
8. Financial services and insurance	6.2	3.07	1.1	1.33	3.3	4.1	4.4	3.40	9.3	7.76	2.2	3.04
9. Public administration and defense	1.5	0.53	0.9	0.63	3.1	3.0	2.5	1.97	4.1	1.46	1.3	1.44
10. Real estate, renting and business activities	4.5	2.05	1.2	1.22	1.7	2.2	5.1	3.67	4.7	3.37	1.3	1.76
11. Education	2.6	1.57	1.3	1.57	1.4	1.8	1.6	1.67	4.4	3.16	0.9	2.77
12. Health and social work	2.4	1.60	1.1	1.20	1.3	2.3	2.2	2.89	6.2	6.18	1.1	1.78
13. Other services activities	0.6	0.50	0.6	0.96	0.6	0.6	2.6	2.89	2.1	2.24	1.2	2.55
14. Private households services	0.5	0.62	0.0	0.09	0.0	1.8	0.0	3.10	6.2	12.24	0.8	5.09
15. Energy average (weighted by LFP)	6.6	4.57	1.9	2.54	3.1	5.3	3.3	3.66	6.5	5.78	2.4	5.23

Table B.6. Relative Sectoral Wages and Sector-Specific Elasticity of Human Capital Accumulation to Schooling Years

Country	Data Sample Size	Data Estimate of Variation Coefficient	Model Estimate of Variation Coefficient	Fréchet Parameter )		
Brazil	8,241,143	6.37	6.37	2.32		
Canada	463,677	1.08	1.08	2.72		
China	24,915	0.91	0.91	2.72		
India	85,855	0.90	0.90	2.80		
Mexico	3,056,419	4.32	4.32	2.33		
United States	1,488,316	1.41	1.41	2.60		

Table B.7. Fréchet Parameter and Variation Coefficient of Wages by Country

Sector		Brazil	(	Canada	China			India		Mexico		ed States
	LFP	LFP	LFP	LFP	LFP	LFP	LFP	LFP	LFP	LFP	LFP	
	Data	Model	Data	Model	Data	Model	Data	Model	Data	Model	Data	Model
		Untargeted		Untargeted		Untargeted		Untargeted		Untargeted		Untargeted
1. Agriculture, hunting, forestry and fishing	10.2	7.0	1.2	0.9	23.8	14.1	33.2	27.9	12.4	6.5	1.0	0.7
2. Manufacturing	11.9	15.7	11.5	13.1	19.6	40.5	14.4	27.5	16.0	32.5	8.7	11.7
3. Water supply	0.5	0.2	0.2	0.1	0.0	0.0	0.1	0.0	0.4	0.1	0.3	0.1
4. Construction	9.6	9.7	7.4	6.4	8.4	6.8	16.7	9.9	16.4	8.7	4.3	2.1
5. Wholesale and retail trade	16.8	18.5	17.6	15.5	11.2	15.8	5.7	6.9	14.0	19.9	15.1	17.0
6. Transport, storage and communications	5.3	5.7	7.3	7.8	3.9	3.7	7.6	10.4	7.6	5.3	7.1	7.4
7. Hotels and restaurants	4.6	2.8	7.0	3.0	2.6	0.9	1.4	0.7	3.5	1.3	8.7	4.4
8. Financial services and insurance	1.3	1.6	4.5	4.4	2.0	2.1	1.5	1.9	1.5	0.4	4.0	3.4
9. Public administration and defense	9.3	23.8	10.8	26.1	3.2	4.5	4.2	5.5	2.1	18.4	14.6	29.7
10. Real estate, renting and business activities	5.9	4.3	8.4	7.6	5.4	2.5	4.6	2.0	9.2	1.8	16.7	13.4
11. Education	7.0	3.4	7.7	3.6	5.3	2.4	6.1	3.4	9.0	1.9	2.3	0.6
12. Health and social work	4.4	2.5	9.2	5.4	3.0	1.1	1.6	0.6	3.5	0.4	12.5	7.2
13. Other services activities	5.6	2.7	4.9	2.2	8.9	3.5	1.4	1.1	2.2	1.1	3.8	1.6
14. Private households services	7.1	1.5	-		0.0	0.0	0.0	0.0	0.8	0.0	0.2	0.0
15. Oil energy production	0.2	0.2	1.1	1.9	0.4	0.3	0.3	0.6	0.7	0.8	0.3	0.2
16. Coal energy production	0.0	0.0	0.1	0.2	1.4	0.9	0.5	0.6	0.1	0.1	0.2	0.1
17. Natural gas energy production	0.0	0.0	0.4	0.8	0.1	0.1	0.1	0.1	0.4	0.4	0.2	0.2
18. Green energy production	0.3	0.4	0.6	1.1	0.7	0.8	0.5	0.8	0.3	0.4	0.3	0.2
Model Fit (Correlation between actual and model LFP series)		80.9%		78.6%		80.2%		89.1%		65.6%		85.3%

Table B.8. Untargeted Labor Force Participation Moment by Country

29

Country	Data	Year	Source
Brazil	Input Output Table	2014	WIOD
	Environmental Accounts to get energy input mix by sector	2009	WIOD
	CO_2 Emissions	2009	WIOD
	Sectoral Labor Force Participation	2014	WIOD
	Sectoral Labor Compensation	2014	WIOD
	Data on Income Earned	2010	IPUMS
	Education Attainment by Sector	2010	IPUMS
	Tatal Labor Free Darticipation (% of GDP)	2015	WDI
	Total Labor Force Participation Rate (%)	2019	WDI
Canada	Input Output Table	2014	WIOD
	Environmental Accounts to get energy input mix by sector	2009	WIOD
	CO_2 Emissions	2009	WIOD
	Sectoral Labor Force Participation	2014	WIOD
	Data on Income Earned	2014	IDLIMS
	Education Attainment by Sector	2011	IPLIMS
	Public Expenditure on Education (% of GDP)	2011	WDI
	Total Labor Force Participation Rate (%)	2011	WDI
China	Innut Output Table	2014	WIOD
China	Environmental Accounts to get energy input mix by sector	2014	WIOD
	CO 2 Emissions	2009	WIOD
	Sectoral Labor Force Participation	2009	WIOD
	Sectoral Labor Compensation	2014	WIOD
	Data on Income Earned	2013	CHIP
	Education Attainment by Sector	2013	CHIP
	Public Expenditure on Education (% of GDP)	2019	MoF China
	Total Labor Force Participation Rate (%)	2019	WDI
India	Input Output Table	2014	WIOD
	Environmental Accounts to get energy input mix by sector	2009	WIOD
	CO_2 Emissions	2009	WIOD
	Sectoral Labor Force Participation	2014	WIOD
	Sectoral Labor Compensation	2014	WIOD
	Data on Wage Earned	2004	IPUMS
	Education Attainment by Sector	2004	IPUMS
	Public Expenditure on Education (% of GDP)	2013	WDI
	Total Labor Force Participation Rate (%)	2019	WDI
Mexico	Input Output Table	2014	WIOD
	Environmental Accounts to get energy input mix by sector	2009	WIOD
	CO_2 Emissions	2009	WIOD
	Sectoral Labor Force Participation	2014	WIOD
	Sectoral Labor Compensation	2014	WIOD
	Education Attainment by Sector	2015	IPUIVIS
	Public Expenditure on Education (% of CDP)	2015	
	Total Labor Force Participation Rate (%)	2010	WDI
		2013	1100
United States	Input Output Table	2014	WIOD
	CO 2 Emissions	2009	WIOD
	Sectoral Labor Force Participation	2009	WIOD
	Sectoral Labor Compensation	2014	WIOD
	Data on Income Earned	2015	IPUMS
	Education Attainment by Sector	2015	IPUMS
	Public Expenditure on Education (% of GDP)	2014	WDI
	Total Labor Force Participation Rate (%)	2019	WDI

Table B.9. Data Sources by Country

Ap	pendix	<b>C</b> :	More	Detailed	Resu	lts
----	--------	------------	------	----------	------	-----

Wasteful Spe	nding Scena	rio						
	%Δ Oil	$\%\Delta$ Coal	$\%\Delta$ Natural Gas	$\%\Delta$ Green	$\%\Delta$ Non-energy	$\%\Delta$ Total Fossil Fuel	$\%\Delta$ Total	Unconditional Target Stated in Nationally Determined Contribution
	Emissions	Emissions	Emissions	Emissions	Emissions	Emissions	Emissions	
Brazil	-27.7%	-26.6%	-26.7%	NaN	-0.4%	-27.5%	-25.9%	37% below 2005 by 2025, 43% by 2030 (indicative)
Canada	-27.9%	-25.4%	-25.1%	NaN	-1.2%	-26.6%	-24.0%	30% below 2005 levels by 2030 (unchanged from NDC)
China	-29.5%	-35.0%	-28.8%	NaN	-2.6%	-34.0%	-32.1%	60-65% carbon intensity reduction by 2030
India	-28.3%	-30.7%	-27.6%	NaN	-1.0%	-29.9%	-28.2%	33 to 35% carbon intensity reduction over 2005 levels by 2030
Mexico	-27.8%	-24.4%	-25.0%	NaN	-1.0%	-26.8%	-25.3%	25% below BAU by 2030 (22% of GHG and a reduction of 51% of Black Carbon).
United States	-27.7%	-26.6%	-25.3%	NaN	-0.7%	-26.8%	-26.0%	26-28% below 2005 levels by 2025
Green Subsid	v Scenario					1		
	%Δ Oil	$\%\Delta$ Coal	$\%\Delta$ Natural Gas	$\%\Delta$ Green	$\%\Delta$ Non-energy	%∆ Total Fossil Fuel	%∆ Total	Unconditional Target Stated in Nationally Determined Contribution
	Emissions	Emissions	Emissions	Emissions	Emissions	Emissions	Emissions	
Brazil	-27.3%	-22.8%	-23.5%	NaN	0.0%	-26.5%	-25.0%	37% below 2005 by 2025, 43% by 2030 (indicative)
Canada	-27.3%	-23.2%	-23.7%	NaN	0.0%	-25.5%	-22.9%	30% below 2005 levels by 2030 (unchanged from NDC)
China	-28.1%	-27.5%	-25.6%	NaN	-1.0%	-27.5%	-25.9%	60-65% carbon intensity reduction by 2030
India	-27.8%	-26.7%	-25.5%	NaN	-0.5%	-26.9%	-25.4%	33 to 35% carbon intensity reduction over 2005 levels by 2030
Mexico	-27.4%	-23.2%	-23.8%	NaN	-0.1%	-26.1%	-24.6%	25% below BAU by 2030 (22% of GHG and a reduction of 51% of Black Carbon).
United States	-27.3%	-23.1%	-23.7%	NaN	0.0%	-25.0%	-24.3%	26-28% below 2005 levels by 2025
Useful Spend	ing Scenario							1
	%∆ Oil	$\%\Delta$ Coal	$\%\Delta$ Natural Gas	$\%\Delta$ Green	$\%\Delta$ Non-energy	%∆ Total Fossil Fuel	%∆ Total	Unconditional Target Stated in Nationally Determined Contribution
	Emissions	Emissions	Emissions	Emissions	Emissions	Emissions	Emissions	
Brazil	-27.3%	-25.8%	-26.0%	NaN	0.2%	-27.1%	-25.5%	37% below 2005 by 2025, 43% by 2030 (indicative)
Canada	-26.3%	-24.9%	-24.2%	NaN	0.8%	-25.4%	-22.8%	30% below 2005 levels by 2030 (unchanged from NDC)
China	-27.2%	-31.9%	-25.5%	NaN	0.5%	-31.0%	-29.1%	60-65% carbon intensity reduction by 2030
India	-26.8%	-29.1%	-25.9%	NaN	0.1%	-28.3%	-26.7%	33 to 35% carbon intensity reduction over 2005 levels by 2030
Mexico	-26.6%	-24.2%	-23.9%	NaN	0.7%	-25.7%	-24.2%	25% below BAU by 2030 (22% of GHG and a reduction of 51% of Black Carbon).
United States	-26.9%	-26.3%	-24.3%	NaN	0.1%	-26.1%	-25.3%	26-28% below 2005 levels by 2025
Education Su	bsidy Scenar	io		-		1		
	%Δ Oil	$\%\Delta$ Coal	$\%\Delta$ Natural Gas	$\%\Delta$ Green	$\%\Delta$ Non-energy	%∆ Total Fossil Fuel	$\%\Delta$ Total	Unconditional Target Stated in Nationally Determined Contribution
	Emissions	Emissions	Emissions	Emissions	Emissions	Emissions	Emissions	, , , , , , , , , , , , , , , , , , ,
Brazil	-27.7%	-26.6%	-26.7%	NaN	-0.4%	-27.5%	-25.9%	37% below 2005 by 2025, 43% by 2030 (indicative)
Canada	-27.9%	-25.4%	-25.1%	NaN	-1.2%	-26.6%	-24.0%	30% below 2005 levels by 2030 (unchanged from NDC)
China	-29.5%	-35.0%	-28.8%	NaN	-2.6%	-34.0%	-32.1%	60-65% carbon intensity reduction by 2030
India	-28.3%	-30.7%	-27.6%	NaN	-1.0%	-29.9%	-28.2%	33 to 35% carbon intensity reduction over 2005 levels by 2030
Mexico	-27.8%	-24.4%	-25.0%	NaN	-1.0%	-26.8%	-25.3%	25% below BAU by 2030 (22% of GHG and a reduction of 51% of Black Carbon).
I Index of Constant	27 70/	26.6%	25.3%	NaN	0.7%	26.8%	26.0%	1.26.28% below 2005 levels by 2025

Table C.1. Percentage Change in  $CO_2$  Emissions by Source, Country and Recycling Scheme

		Br	azil			Car	ada			Ch	ina			In	dia			Me	xico			United	States	
	Sales	VA	Int.	LFP	Sales	VA	Int.	LFP																
			Cons.				Cons.				Cons.				Cons.				Cons.				Cons.	
1. Agriculture, hunting, forestry and fishing	4.5%	5.2%	3.7%	10.2%	2.3%	1.6%	3.1%	1.2%	5.3%	9.4%	3.2%	23.8%	10.4%	14.8%	5.5%	33.2%	3.0%	3.3%	2.7%	12.4%	1.6%	1.2%	2.0%	1.0%
<ol><li>Manufacturing</li></ol>	27.6%	14.6%	43.6%	11.9%	18.9%	11.6%	26.9%	11.5%	50.0%	30.1%	59.7%	19.6%	36.4%	16.6%	58.5%	14.4%	34.4%	18.8%	55.6%	16.0%	20.1%	12.4%	29.9%	8.7%
3. Water supply	0.6%	0.7%	0.4%	0.5%	0.2%	0.3%	0.1%	0.2%	0.2%	0.3%	0.2%	0.0%	0.2%	0.2%	0.2%	0.1%	0.4%	0.4%	0.3%	0.4%	0.3%	0.3%	0.4%	0.3%
4. Construction	7.2%	6.7%	7.8%	9.6%	8.9%	7.7%	10.3%	7.4%	9.6%	6.8%	10.9%	8.4%	9.9%	7.2%	12.8%	16.7%	7.8%	7.6%	8.1%	16.4%	3.9%	3.8%	4.0%	4.3%
5. Wholesale and retail trade	10.3%	12.4%	7.6%	16.8%	10.3%	10.5%	10.1%	17.6%	5.3%	9.7%	3.1%	11.2%	10.6%	17.1%	3.3%	5.7%	12.5%	16.8%	6.7%	14.0%	10.5%	12.2%	8.4%	15.1%
6. Transport, storage and communications	8.4%	8.0%	8.8%	5.3%	9.1%	8.0%	10.3%	7.3%	4.8%	7.2%	3.6%	3.9%	10.4%	11.6%	9.0%	7.6%	8.3%	7.9%	8.9%	7.6%	9.5%	9.1%	9.9%	7.1%
7. Hotels and restaurants	2.3%	2.4%	2.3%	4.6%	2.2%	2.1%	2.3%	7.0%	1.7%	1.9%	1.6%	2.6%	2.2%	1.4%	3.1%	1.4%	1.9%	2.2%	1.4%	3.5%	2.9%	2.8%	3.0%	8.7%
8. Financial services and insurance	5.5%	6.3%	4.5%	1.3%	5.6%	5.5%	5.8%	4.5%	2.9%	6.0%	1.3%	2.0%	3.4%	5.5%	1.1%	1.5%	3.1%	3.5%	2.6%	1.5%	7.0%	7.0%	7.1%	4.0%
9. Public administration and defense	11.7%	16.6%	5.7%	9.3%	15.9%	19.5%	12.0%	10.8%	5.6%	9.7%	3.7%	3.2%	4.8%	7.7%	1.6%	4.2%	11.8%	17.8%	3.7%	2.1%	19.4%	23.1%	14.7%	14.6%
10. Real estate, renting and business activities	7.2%	9.6%	4.2%	5.9%	8.6%	9.0%	8.1%	8.4%	2.4%	4.0%	1.6%	5.4%	3.5%	6.7%	0.0%	4.6%	3.7%	4.4%	2.7%	9.2%	11.1%	13.1%	8.5%	16.7%
11. Education	3.9%	5.5%	1.8%	7.0%	3.5%	5.4%	1.4%	7.7%	1.9%	3.3%	1.2%	5.3%	2.3%	4.0%	0.5%	6.1%	2.8%	4.3%	0.8%	9.0%	1.0%	1.1%	0.9%	2.3%
12. Health and social work	3.5%	4.2%	2.6%	4.4%	4.6%	6.3%	2.9%	9.2%	1.7%	1.8%	1.7%	3.0%	1.2%	1.5%	0.9%	1.6%	2.0%	2.3%	1.5%	3.5%	6.7%	7.1%	6.3%	12.5%
<ol><li>Other services activities</li></ol>	1.8%	1.8%	1.9%	5.6%	2.0%	2.1%	1.9%	4.9%	1.6%	2.3%	1.3%	8.9%	1.6%	2.4%	0.6%	1.4%	1.3%	1.6%	0.9%	2.2%	2.4%	2.6%	2.3%	3.8%
<ol><li>Private households services</li></ol>	0.6%	1.1%	0.0%	7.1%	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%	0.0%	0.3%	0.5%	0.0%	0.8%	0.1%	0.1%	0.1%	0.2%
15. Oil energy production	1.5%	1.7%	1.3%	0.2%	3.7%	4.9%	2.3%	1.1%	1.0%	1.2%	0.8%	0.4%	0.6%	0.6%	0.5%	0.3%	3.2%	4.3%	1.8%	0.7%	1.0%	1.3%	0.6%	0.3%
16. Coal energy production	0.1%	0.1%	0.1%	0.0%	0.4%	0.5%	0.3%	0.1%	3.8%	2.8%	4.3%	1.4%	1.2%	0.9%	1.5%	0.5%	0.3%	0.3%	0.2%	0.1%	0.7%	0.7%	0.6%	0.2%
<ol><li>Natural gas energy production</li></ol>	0.4%	0.3%	0.4%	0.0%	1.5%	2.1%	0.9%	0.4%	0.3%	0.3%	0.3%	0.1%	0.2%	0.1%	0.2%	0.1%	1.7%	2.0%	1.3%	0.4%	0.8%	0.9%	0.5%	0.2%
18. Green energy production	2.9%	2.9%	3.0%	0.3%	2.1%	2.9%	1.3%	0.6%	2.0%	3.2%	1.4%	0.7%	1.2%	1.6%	0.7%	0.5%	1.5%	2.0%	0.9%	0.3%	1.0%	1.2%	0.7%	0.3%
Sum of total dirty energy production shares	2.0%	2.0%	1.9%	0.2%	5.6%	7.5%	3.5%	1.7%	5.1%	4.4%	5.4%	1.9%	1.9%	1.7%	2.1%	0.9%	5.2%	6.6%	3.3%	1.1%	2.4%	3.0%	1.8%	0.6%

Table C.2. Sectoral Breakdown of Output, VA, Intermediate Consumption and Labor Force Participation by Country



 $\label{eq:Figure C.1: Percentage Change in Real Consumption and Leisure for Each Decile.$ 

	Relativ	e Wages	LI	FP
	Data	Model	Data	Model
1. Agriculture, hunting, forestry and fishing	1.00	1.00	12.4%	14.4%
2. Manufacturing	2.49	2.49	16.0%	17.2%
3. Water supply	2.66	2.66	0.4%	0.3%
4. Construction	2.61	2.61	16.4%	18.2%
5. Wholesale and retail trade	3.54	3.54	14.0%	15.4%
6. Transport, storage and communications	4.24	4.24	7.6%	7.1%
7. Hotels and restaurants	2.73	2.73	3.5%	3.2%
8. Financial services and insurance	9.26	9.26	1.5%	1.0%
9. Public administration and defense	4.14	5.23	2.1%	2.2%
10. Real estate, renting and business activities	4.68	4.68	9.2%	6.9%
11. Education	4.38	4.38	9.0%	7.2%
12. Health and social work	6.23	6.23	3.5%	1.4%
13. Other services activities	2.11	2.11	2.2%	2.5%
14. Private households services	6.20	6.20	0.8%	0.1%
15. Oil energy production	6.46	6.50	0.7%	1.4%
16. Coal energy production			0.1%	0.1%
17. Natural gas energy production			0.4%	0.8%
18. Green energy production			0.3%	0.7%
Correlation	99	9.2%	98.	3%

Table D.1. Model Fit Upon Introducing Wage-Wedges in Mexico

Mexico - with wedges	GDP	Consumption	Cons. Equiv.
Wasteful Spending	-1.1	-3.3	-2.4
Green Subsidy	-0.7	-0.7	-0.8
Useful Spending	-0.9	-0.9	0.5
Education Subsidy	1.0	-1.3	-0.3
Mexico - without wedges	GDP	Consumption	Cons. Equiv.
<b>Mexico</b> - <b>without wedges</b> Wasteful Spending	<b>GDP</b> -1.1	Consumption -3.4	Cons. Equiv. -2.2
<b>Mexico - without wedges</b> Wasteful Spending Green Subsidy	<b>GDP</b> -1.1 -0.7	Consumption -3.4 -0.7	<b>Cons. Equiv.</b> -2.2 -0.8
<b>Mexico</b> - <b>without wedges</b> Wasteful Spending Green Subsidy Useful Spending	GDP -1.1 -0.7 -1.0	Consumption -3.4 -0.7 -1.0	<b>Cons. Equiv.</b> -2.2 -0.8 0.4
Mexico - without wedges Wasteful Spending Green Subsidy Useful Spending Education Subsidy	GDP -1.1 -0.7 -1.0 1.0	Consumption -3.4 -0.7 -1.0 -1.4	<b>Cons. Equiv.</b> -2.2 -0.8 0.4 0.0

Table D.2. Comparing Main Results for Mexico With and Without Wage Wedges

#### Appendix D: Introducing Wage-Wedges

In this subsection, we add wage-wedges to the case of Mexico, which had the lowest correlation between the untargeted labor force participation shares in the model and in the data. The model fit with the newly introduced wedges is now presented in table Table D.1.

Results in Table D.2 prove robust to the introduction of wedges showing that leaving the labor force participation shares untargeted in the model has little to no effect on the aforementioned aggregate effects.

## Working Papers

## 2017

- 1|17 The diffusion of knowledge via managers' mobility Giordano Mion | Luca David Opromolla | Alessandro Sforza
- 2 | 17 Upward nominal wage rigidity Paulo Guimarães | Fernando Martins | Pedro Portugal
- 3|17 Zooming the ins and outs of the U.S. unemployment Pedro Portugal | António Rua
- 4|17 Labor market imperfections and the firm's wage setting policy Sónia Félix | Pedro Portugal
- 5|17 International banking and cross-border effects of regulation: lessons from Portugal Diana Bonfim | Sónia Costa
- 6 | 17 Disentangling the channels from birthdate to educational attainment Luís Martins | Manuel Coutinho Pereira
- 7|17 Who's who in global value chains? A weighted network approach

João Amador | Sónia Cabral | Rossana Mastrandrea | Franco Ruzzenenti

 8|17 Lending relationships and the real economy: evidence in the context of the euro area sovereign debt crisis
 Luciana Barbosa

- 9|17 Impact of uncertainty measures on the Portuguese economy Cristina Manteu | Sara Serra
- 10|17 Modelling currency demand in a small open economy within a monetary union António Rua
- 11|17 Boom, slump, sudden stops, recovery, and policy options. Portugal and the Euro Olivier Blanchard | Pedro Portugal
- 12|17 Inefficiency distribution of the European Banking System João Oliveira
- 13|17 Banks' liquidity management and systemic risk Luca G. Deidda | Ettore Panetti
- 14|17 Entrepreneurial risk and diversification through trade Federico Esposito
- 15|17 The portuguese post-2008 period: a narrative from an estimated DSGE model Paulo Júlio | José R. Maria
- 16|17 A theory of government bailouts in a heterogeneous banking systemFilomena Garcia | Ettore Panetti
- 17|17 Goods and factor market integration: a quantitative assessment of the EU enlargement FLorenzo Caliendo | Luca David Opromolla | Fernando Parro | Alessandro Sforza

## 2018

- 1|18 Calibration and the estimation of macroeconomic models Nikolay Iskrev
- 2|18 Are asset price data informative about news shocks? A DSGE perspective Nikolay Iskrev
- 3|18 Sub-optimality of the friedman rule with distorting taxes Bernardino Adão | André C. Silva
- 4 | 18 The effect of firm cash holdings on monetary policy Bernardino Adão | André C. Silva
- 5|18 The returns to schooling unveiled Ana Rute Cardoso | Paulo Guimarães | Pedro Portugal | Hugo Reis
- 6|18 Real effects of financial distress: the role of heterogeneity

Francisco Buera | Sudipto Karmakar

- 7|18 Did recent reforms facilitate EU labour market adjustment? Firm level evidence Mario Izquierdo | Theodora Kosma | Ana Lamo | Fernando Martins | Simon Savsek
- 8|18 Flexible wage components as a source of wage adaptability to shocks: evidence from European firms, 2010–2013

Jan Babecký | Clémence Berson | Ludmila Fadejeva | Ana Lamo | Petra Marotzke | Fernando Martins | Pawel Strzelecki

#### 9|18 The effects of official and unofficial information on tax compliance

Filomena Garcia | Luca David Opromolla Andrea Vezulli | Rafael Marques

- 10|18 International trade in services: evidence for portuguese firms João Amador | Sónia Cabral | Birgitte Ringstad
- 11|18 Fear the walking dead: zombie firms, spillovers and exit barriers Ana Fontoura Gouveia | Christian Osterhold
- 12|18 Collateral Damage? Labour Market Effects of Competing with China – at Home and Abroad

Sónia Cabral | Pedro S. Martins | João Pereira dos Santos | Mariana Tavares

- 13|18 An integrated financial amplifier: The role of defaulted loans and occasionally binding constraints in output fluctuations
   Paulo Júlio | José R. Maria
- 14|18Structural Changes in the Duration of Bull<br/>Markets and Business Cycle Dynamics<br/>João Cruz | João Nicolau | Paulo M.M.<br/>Rodrigues
- 15|18 Cross-border spillovers of monetary policy: what changes during a financial crisis?
   Luciana Barbosa | Diana Bonfim | Sónia Costa | Mary Everett
- 16|18 When losses turn into loans: the cost of undercapitalized banks

Laura Blattner | Luísa Farinha | Francisca Rebelo

17|18 Testing the fractionally integrated hypothesis using M estimation: With an application to stock market volatility

> Matei Demetrescu | Paulo M. M. Rodrigues | Antonio Rubia

- 18|18 Every cloud has a silver lining: Micro-level evidence on the cleansing effects of the Portuguese financial crisisDaniel A. Dias | Carlos Robalo Marques
- 19|18 To ask or not to ask? Collateral versus screening in lending relationships Hans Degryse | Artashes Karapetyan | Sudipto Karmakar
- 20|18 Thirty years of economic growth in Africa João Amador | António R. dos Santos
- 21|18 CEO performance in severe crises: the role of newcomers

Sharmin Sazedj | João Amador | José Tavares

22|18 A general equilibrium theory of occupational choice under optimistic beliefs about entrepreneurial ability Michele Dell'Era | Luca David Opromolla | Luís Santos-Pinto

- 23|18 Exploring the implications of different loanto-value macroprudential policy designs Rita Basto | Sandra Gomes | Diana Lima
- 24|18 Bank shocks and firm performance: new evidence from the sovereign debt crisis Luísa Farinha | Marina-Eliza Spaliara | Serafem Tsoukas
- 25|18 Bank credit allocation and productivity: stylised facts for Portugal Nuno Azevedo | Márcio Mateus | Álvaro Pina
- 26|18 Does domestic demand matter for firms' exports? Paulo Soares Esteves | Miguel Portela | António Rua
- 27|18 Credit Subsidies Isabel Correia | Fiorella De Fiore | Pedro Teles | Oreste Tristani

## 2019

1|19 The transmission of unconventional monetary policy to bank credit supply: evidence from the TLTRO

António Afonso | Joana Sousa-Leite

- 2|19 How responsive are wages to demand within the firm? Evidence from idiosyncratic export demand shocks Andrew Garin | Filipe Silvério
- 3|19 Vocational high school graduate wage gap: the role of cognitive skills and firms Joop Hartog | Pedro Raposo | Hugo Reis

- 4|19 What is the Impact of Increased Business Competition? Sónia Félix | Chiara Maggi
- 5|19 Modelling the Demand for Euro Banknotes António Rua
- 6|19 Testing for Episodic Predictability in Stock Returns

Matei Demetrescu | Iliyan Georgiev Paulo M. M. Rodrigues | A. M. Robert Taylor

7 | 19 The new ESCB methodology for the calculation of cyclically adjusted budget balances: an application to the Portuguese case Cláudia Braz | Maria Manuel Campos Sharmin Sazedj

- 8|19 Into the heterogeneities in the Portuguese labour market: an empirical assessment Fernando Martins | Domingos Seward
- **9|19** A reexamination of inflation persistence dynamics in OECD countries: A new approach

Gabriel Zsurkis | João Nicolau | Paulo M. M. Rodrigues

- 10|19 Euro area fiscal policy changes: stylised features of the past two decades Cláudia Braz | Nicolas Carnots
- 11|19 The Neutrality of Nominal Rates: How Long is the Long Run? João Valle e Azevedo | João Ritto | Pedro

Teles

12|19 Testing for breaks in the cointegrating relationship: on the stability of government bond markets' equilibrium

> Paulo M. M. Rodrigues | Philipp Sibbertsen Michelle Voges

#### 13|19 Monthly Forecasting of GDP with Mixed Frequency MultivariateSingular Spectrum Analysis

Hossein Hassani | António Rua | Emmanuel Sirimal Silva | Dimitrios Thomakos

14|19 ECB, BoE and Fed Monetary-Policy announcements: price and volume effects on European securities markets

Eurico Ferreira | Ana Paula Serra

- 15|19 The financial channels of labor rigidities: evidence from Portugal Edoardo M. Acabbi | Ettore Panetti | Alessandro Sforza
- 16|19 Sovereign exposures in the Portuguese banking system: determinants and dynamics
   Maria Manuel Campos | Ana Rita Mateus | Álvaro Pina
- 17|19 Time vs. Risk Preferences, Bank Liquidity Provision and Financial Fragility Ettore Panetti
- 18|19 Trends and cycles under changing economic conditionsCláudia Duarte | José R. Maria | Sharmin Sazedj
- **19|19** Bank funding and the survival of start-ups Luísa Farinha | Sónia Félix | João A. C. Santos
- 20|19 From micro to macro: a note on the analysis of aggregate productivity dynamics using firm-level data

Daniel A. Dias | Carlos Robalo Marques

21|19 Tighter credit and consumer bankruptcy insurance

António Antunes | Tiago Cavalcanti | Caterina Mendicino | Marcel Peruffo | Anne Villamil

## 2020

- 1 20 On-site inspecting zombie lending Diana Bonfim | Geraldo Cerqueiro | Hans Degryse | Steven Ongena
- 2|20 Labor earnings dynamics in a developing economy with a large informal sector
   Diego B. P. Gomes | Felipe S. lachan | Cezar Santos
- 3|20 Endogenous growth and monetary policy: how do interest-rate feedback rules shape nominal and real transitional dynamics? Pedro Mazeda Gil | Gustavo Iglésias
- 4|20 Types of International Traders and the Network of Capital Participations João Amador | Sónia Cabral | Birgitte Ringstad
- 5|20 Forecasting tourism with targeted predictors in a data-rich environment Nuno Lourenço | Carlos Melo Gouveia | António Rua
- 6|20 The expected time to cross a threshold and its determinants: A simple and flexible framework

Gabriel Zsurkis | João Nicolau | Paulo M. M. Rodrigues

7|20 A non-hierarchical dynamic factor model for three-way data

Francisco Dias | Maximiano Pinheiro | António Rua

8|20 Measuring wage inequality under right censoring João Nicolau | Pedro Raposo | Paulo M. M.

Rodrigues

9|20 Intergenerational wealth inequality: the role of demographics

António Antunes | Valerio Ercolani

- 10|20 Banks' complexity and risk: agency problems and diversification benefits Diana Bonfim | Sónia Felix
- 11|20 The importance of deposit insurance credibility Diana Bonfim | João A. C. Santos
- 12|20 Dream jobs Giordano Mion | Luca David Opromolla | Gianmarco I.P. Ottaviano
- 13|20 The DEI: tracking economic activity daily during the lockdown

Nuno Lourenço | António Rua

- 14|20 An economic model of the Covid-19 pandemic with young and old agents: Behavior, testing and policies Luiz Brotherhood | Philipp Kircher | Cezar Santos | Michèle Tertilt
- 15|20 Slums and Pandemics Luiz Brotherhood | Tiago Cavalcanti | Daniel Da Mata | Cezar Santos
- 16|20 Assessing the Scoreboard of the EU Macroeconomic Imbalances Procedure: (Machine) Learning from Decisions

Tiago Alves | João Amador | Francisco Gonçalves

17|20 Climate Change Mitigation Policies: Aggregate and Distributional Effects Tiago Cavalcanti | Zeina Hasna | Cezar Santos